# Social Vulnerability to Coastal Floods in Denmark:

# investigating the spatial patterns among the Danish municipalities

6/6/2019



Maria Pappa Study no. 20171884



#### Title:

Social Vulnerability to Coastal Floods in Denmark: investigating the spatial patterns among the Danish municipalities

#### **Project period:**

01/02/2019 -07/06/2019

Supervisor:

James Fitton

Page number:

57

Appendix:

I - VIII

**Finished:** 07.06.2019

Author:

Maria Pappa

#### Abstract:

Climate change is expected to exacerbate the occurrence of coastal floods in Denmark. Concerned by the limited research of the social vulnerabilities in the flood risk assessments in Denmark, this research aims at quantifying the social vulnerability of the Danish population. Municipal-level socio-economic data are used to construct a social vulnerability index, based on the SoVI-Lite methodology, a simplified version of SoVI®. The social vulnerability to coastal flood is quantified for 97 Danish municipalities. The majority of the municipalities exhibit moderate levels of vulnerability. Twenty municipalities scored high levels of vulnerability and generate two clusters; one containing the four most populated and urban municipalities, and one with sparsely populated rural islands in the inner coast of the country. The municipalities with the lowest level of vulnerability to coastal floods tend to cluster around Copenhagen municipality. With the use of factor analysis three components of social vulnerability are identified as the most influential in the construction of vulnerability to floods. These factors explain 83% of the data variance. The findings suggest that vulnerability is connected with the particular socio-demographic challenges of each municipality. The findings can be incorporated in the existing flood risk assessments, while they can the work of municipalities in support the implementation of local flood management plans and raise awareness of the vulnerabilities to coastal flooding in the Danish society.

The reports content is available for everyone but publishing it (with sources) may only happen by approval of the authors

## Acknowledgements

The research was conducted as a 10<sup>th</sup> semester master thesis under the Master's Program in Geography of Aalborg University. I would like to express my gratitude to my supervisor, Dr. James Fitton, for his valuable help and guidance during this process of this research. For this, I thank James for his patience and support during this thesis. I would like to thank my interviewee, Ms. Ulla Varneskov, for her insights regarding the topic of the research.

Finally, a special thank you note to our project coordinator, Mrs. Carla Smink, as well as Mr. Lars Bodum and Mrs. Paulina Ramirez-Monsalve, for their overall academic support and their tremendous help during my time at Aalborg University.

## **Table of Contents**

Ab	stract	t	1
Ac	know	vledgements	2
Lis	t of T	Tables	5
Lis	t of F	Figures	6
List of Abbreviations			7
1	Intr	roduction	8
	1.1	Climate Facts	8
	1.2	Construction of Disasters	9
	1.3	Floods and the associated impacts	9
	1.4	Coastal Floods in Europe and Denmark	11
	1.5	Legislation and initiatives	12
	1.6	Relevance of study	13
	1.7	Research questions	15
	1.8	Thesis Outline	15
2	Res	search Design	16
4	2.1	Theory of science	16
	2.1.	.1 Constructivism	16
4	2.2	Methods	17
	2.2.	2.1 Qualitative Methods	17
	2.2.	2.2 Quantitative Methods	18
	2.2.		
	2.2.	Principal Component Analysis (PCA)	19
4	2.3	Limitations	19
	2.3.	3.1 Limitations	19
	2.3.	B.2 Delimitations	19
3	The	eoretical framework – Key Concepts	20
	3.1	Risk	20
	3.2	Vulnerability	21
	3.2.	2.1 Vulnerability models	23
	3.2.	Hazard of place model	25

3	.3	Exposure	27
3	.4	Susceptibility	27
3	.5	Glossary	
4	Dev	eloping the Social Vulnerability Index	
4	.1	Social Vulnerability Indicators	
4	.2	Social Vulnerability Data	
4	.3	Building the social vulnerability index	
5	Resu	ults	
5	.1	Where do vulnerable people live?	
5	.2	Which are the main factors of social vulnerability?	
6	Disc	cussion	42
6	.1	Key findings	42
6	.2	Implications for flood risk management in Denmark	43
6	.3	Limitations of the study	
6	.4	Future research recommendations	
7	Con	clusion	46
8	Bibl	liography	

## List of Tables

# List of Figures

Figure 1.1: Flooding events across the word from 1986 - 201510
Figure 1.2: Number of coastal floods and number of reported damages from coastal floods per municipality,
from 1991 – 2009 (Stormrådet, 2009)
Figure 1.3: Areas at risk of flooding, as identified for the implementation of the EU Flood Directive. The areas
with red are the ones identified through the preliminary risk assessment, while the blue ones represent the areas
identified by the second assessment (Kystdirektoratet, 2018b)
Figure 1.4: Hazard map of potential coastal flooding in Denmark in the years 2015 and 2115 (Jebens &
Sorensen, 2017)
Figure 3.1 The widening of vulnerability concept, adapted from Birkmann (2005b; 2006)23
Figure 3.2: Pressure and Release Model (Wisner et al., 2004)
Figure 3.3: The hazard-of-place model (Cutter et al., 2000; Cutter, 1996; Cutter et al., 2003). This research aims
at investigating the "social fabric" in Denmark in order to accomplish measuring social vulnerability26
Figure 5.1: Social vulnerability of Danish municipalities. The municipalities around the metropolitan area of
Copenhagen are enlarged to provide a detailed view due to the concentration of small municipalities The
municipalities are ranked based on the standard deviation of the derived vulnerability scores, and are classified
according to the legend of the map
Figure 5.2: Scree plot diagram of the derived principal components. The three components have eigenvalues
greater than +1.0 and are the components that contains the greatest amount of information of the original
indicator values. The three components are named Age, gender and socioeconomic status, Immigrants,
indicator values. The three components are named Age, gender and socioeconomic status, Immigrants, Urbanism and special needs, and Accessibility and ethnicity. The individual indicators of each component are
Urbanism and special needs, and Accessibility and ethnicity. The individual indicators of each component are explained further within the chapter
Urbanism and special needs, and Accessibility and ethnicity. The individual indicators of each component are explained further within the chapter
Urbanism and special needs, and Accessibility and ethnicity. The individual indicators of each component are explained further within the chapter
Urbanism and special needs, and Accessibility and ethnicity. The individual indicators of each component are explained further within the chapter
Urbanism and special needs, and Accessibility and ethnicity. The individual indicators of each component are explained further within the chapter
Urbanism and special needs, and Accessibility and ethnicity. The individual indicators of each component are explained further within the chapter
Urbanism and special needs, and Accessibility and ethnicity. The individual indicators of each component are explained further within the chapter
Urbanism and special needs, and Accessibility and ethnicity. The individual indicators of each component are explained further within the chapter
Urbanism and special needs, and Accessibility and ethnicity. The individual indicators of each component are explained further within the chapter

## List of Abbreviations

CCA - Climate Change Adaptation DCA – Danish Coastal Authority DRM - Disaster Risk Management GIS - Geographic Information Systems IPCC - the Intergovernmental Panel on Climate Change LECZ - Low Elevation Coastal Zones PAR - Pressure and release SoVI – Social Vulnerability Index SLR - Sea Level Rise UN - United Nations UNDP - United Nations Development Programme UNDRR - United Nations Office for Disaster Risk Reduction UNEP - United Nations Environmental Programme UNISDR - United Nations Strategy for Disaster Risk Reduction

## **1** Introduction

Past flooding disasters have forced local and global authorities to launch and implement risk reduction strategies. Among the tools for accomplishing this goal, vulnerability assessments hold a prominent position. Vulnerability in general is the propensity to harm due to the exposure to different risks, hazards or stressors (Birkmann et al., 2013; Cutter & Emrich, 2006; Villagran De Leon, 2006). Vulnerability comprises an indispensable component in the risk reduction process across the world, because it has been proved that for successful disaster preparedness knowledge of the social drivers of vulnerability is crucial (Marchand, 2009). Therefore, different tools have been developed in order to quantify vulnerability in a specific context. As the climate continues to change and the occurrence of natural hazards becomes more frequent, identification of vulnerable population can be a tool for the governments and planners. As Birkmann (2005) points out risk reduction should also focus on identification and assessment of the vulnerabilities of people, instead of just focusing on the type of hazard. Concerned by the future risk of coastal flooding in Denmark due to the local impacts of climate change, this study will focus on identifying the population with the biggest propensity to harm in case of coastal flooding, the spatial distribution of social vulnerabilities and the potential spatial clusters of it.

#### 1.1 Climate Facts

Climate change is a recognised discourse that has emerged over the past decades, with 97% of the published peer-reviewed climate research endorsing that the climate warming trends of the last decades are caused or accelerated by human activities (Cook et al., 2016). The evidence of the changing climate are undisputable; mean temperatures are rising, oceans are getting warmer, ice sea caps are shrinking exponentially, the sea levels are also rising, extreme weather events are becoming more frequent, are only some of the evidence globally (Nerem et al., 2018; Sun et al., 2006; U.S. Global Change Research Program, 2017; Wuebbles et al., 2017). Another agreement in the scientific community is that the climate will continue changing in the future, exacerbating the undesirable effects for human and ecological systems (EEA, 2017; Vinod, 2017).

The climate projections are alarming by verifying the aggravation of the climate change effects and the occurrence of more natural hazards (Cook et al., 2016; Doran & Zimmerman, 2009; Oreskes, 2004). Indeed, natural hazards related with hydro-meteorological events, such as floods and storms, have been increased exponentially in recent years, with tremendous human and economic losses (Vinod, 2017). UNEP (2015) estimates that the expected cost for adaptation for developing countries in billion US\$ will increase dramatically from 150-300 billion in 230 up to 280-500 billion in 2050. The increase in climate adaptation cost globally is also supported by the findings of Lipinski and McGray (2010) who summarise existing estimations of the expected adaptation cost and point out an increase of tens of billions annually.

Coastal areas are considered especially vulnerable to the different impacts of climate change (McGranahan, Balk, & Anderson, 2007; Nicholls, Hoozemans, & Marchand, 1999; Sorensen et al., 2018) Disasters like Hurricane Katrina, point out that the coastal areas are prone to hazards (Marchand, 2009). Furthermore, the high concentration of human settlements in the coastal areas, due to the socio-economic assets of the coast, creates additional pressure in the increased exposure to natural hazards (Marchand, 2009). Hauer et al. (2016) conclude that with an average sea level rise (SLR) of 0.9 m up until 2100, just in the USA alone, 4.2 millions of people will be at risk of the destructive impacts of high sea levels. SLR is a common threat for coastal areas with low

elevation zones, like Denmark. Denmark is ranked in the top ten countries with the largest share of land in the low elevation coastal zone, with a share of 26% (Table 1.1). It is not surprising the fact the largest urban centers of the country are located near the coast and majority of the population lives in close proximity to the coast (Jebens & Sorensen, 2017). As a result, the expected impacts of climate change in Denmark increase the risk posed by the rising seas, and potentially could have substantial impacts in the socioeconomic activity of the country as a whole.

Table 1.1: Classification of countries with the largest amount of land in the Low Elevation Coastal Zone (LECZ). Areas along the coast that are located 10m below the sea level are classified as LECZ. Even though LECZ cover only 2% of the earth's surface, 10% of the global population resides in these zones (McGranahan et al., 2007)

Ranked by share of their land in the LECZ			
		Land area in LECZ	
Country**	Overall rank*	('000 km²)	%
Bahamas	158	12	93
Netherlands	132	31	75
Bangladesh	93	54	40
French Polynesia	171	1	32
The Gambia	163	3	26
Denmark	131	11	26
Qatar	162	3	23
Cuba	102	23	21
Vietnam	65	66	20
Guinea-Bissau	137	7	19

#### **1.2** Construction of Disasters

Wisner et al. (2004) suggest that natural disasters are never truly natural. Thus, disasters encompass a natural and a social aspect. A disaster might be triggered by natural events, such as flooding or hurricanes, but the way it affects people and different areas is defined by the social, economic and power structures that shape the life of those people (Wisner et al., 2004). Therefore, is it important to recognize the importance of social, economic and demographic characteristics that can amplify or reduce the impacts of a natural hazard. The toll of these events is unequal for people around the world, and the most vulnerable people will suffer the most.

A disaster that pointed out the relation between social vulnerability and the distribution of the impacts of a natural hazard was hurricane Katrina, that stroke on August, 2005 in the Louisiana-Mississippi border (Myers et al., 2008). According to the findings of Myers et al. (2008), the biggest cost of the hurricane was distributed among the poorest population, dwellers of old buildings and members of ethnic minorities. Cutter and Emrich (2006) indicate that people with disabilities, elderly population, the population with lower than the average annual income and immigrants were identified as the most vulnerable in the case of hurricane Katrina.

#### **1.3** Floods and the associated impacts

Globally millions of lives are affected, altered or destroyed by natural hazards every year (Vinod, 2017). Among the different types of hazards, a prominent position is held by floods. Kundzweicz (2012) defines flood as

"temporal presence of water in normally dry areas". Floods are occurring all over the world (Figure 1.1), manifested in different types, fluvial floods due to river overflow, flash floods due to intense precipitation, coastal surges due to storms that elevate sea water, or drainage overflow especially in urban areas (Rufat et al., 2015). The number of flood events from 1986 until 2005 is illustrated in Figure 1.1, which shows that flooding constitute a threat for people both in developed and developing countries, without many exceptions.

Regardless the type of the flood, the impacts can be crucial and even long-lasting in human lives, thus floods are considered destructive across the globe (Chorynski et al., 2012; Kundzewicz, 2012; Rodda et al., 2012). Floods are characterised by multi-causality, thus their impacts are determined not only by climatic factors. As Kundzweicz (2012) further states multiple socio-economic characteristics, like economic status and population density, affect the impacts of floods.



Figure 1.1: Flooding events across the word from 1986 – 2015 (CRED, 2015)

One could argue that the consequences of floods are bigger in less developed countries. Multiple flood disasters in the Philippines, Bangladesh and Pakistan illustrate the importance of the non-climatic factors that affect the flood consequences (Kundzewicz et al., 2014). In general, wealthy countries develop risk management plans and own warning systems, which can prevent or minimize human losses, while it is possible to reduce the material losses after a disaster (Peduzzi et al., 2013; Rodda et al., 2012; Vinod, 2017). However, that's not always the case; in the USA and Europe despite the constant efforts and investments in flood protection measures, the economic damages caused by floods hazards have not decreased, and floods still constitute a major threat (Mostert & Junier, 2010; Peduzzi et al., 2013). Globally, the material cost of flooding in the recent decades exceeds tens of billions of US\$ every year (Kundzewicz, 2012; Kundzewicz et al., 2014; Peduzzi et al., 2013).

More precisely, the increase of flood hazards in Europe in the last decades has significantly raised the cost of economic damages up to 52 billion euros for flood related losses and 44 billion euros for storm related damages (European Environment Agency, 2011; Peduzzi et al., 2013).

#### 1.4 Coastal Floods in Europe and Denmark

During the last 50 years, the economic losses associated with floods have been increased in the entire world (Chorynski et al., 2012; Grothmann & Reusswig, 2006; Kundzewicz et al., 2014). EU has experienced severe economic and even human losses; some examples of member states that have experienced flood disasters in the 20th century are Italy, Spain, Romania, Ukraine an Poland (Chorynski et al., 2012; Peduzzi et al., 2013).

In Denmark the biggest flood event was the large flood of November 1872 (Jebens & Sorensen, 2017). Jebens and Sorensen (2017) argue that this event was the worst coastal flood Denmark has ever experienced, with 250 human losses and 15,000 homeless people around the Southwest Baltic Sea. In addition, Denmark has been hit by multiple storm surges, especially in the last two decades (Jebens & Sorensen, 2017). Moreover, a major part of the Danish coastline has been hit by a storm surge at least one time between the years 1991 and 2009 (Figure 1.2). Moreover, the number of the reported damages per municipality for the same period is illustrated in Figure 1.2. According to the Danish Storm Council (2009), from 1991 until 2009, they provided almost 550 billion DDK for compensation for coastal flooding and storm surges material damages, and they conclude that the worst storm surge was the one of November 2006 with almost 4,000 reported material damages incidents (Figure 1.2).



Figure 1.2: Number of coastal floods and number of reported damages from coastal floods per municipality, from 1991 – 2009 (Stormrådet, 2009)

The fluvial risk, because of the absence of big riverine systems, in Denmark is considered limited (Jebens et al., 2016). Jebens and Sorensen (2017) further claim that the risk of pluvial floods is relatively low as well. Meanwhile, notwithstanding this claim, in the last decade Denmark has experienced floods from extreme precipitation in many urban areas with significant economic damages. An example of how impactful the changing climate can be for the Danish society is the extreme rainfall in the summer of 2011, which result in sudden flood in Copenhagen with overall material losses of 6 billion DKK (Danish Government, 2012). This event was an alarming call for the Danish Government and the related stakeholders, highlighting that extreme weather and hydrological events are sudden and can disturb the everyday life of the Danish population.

The Danish Task Force on Climate Change Adaptation identified as the main challenges for Denmark the warmer summers, more extreme weather events, and changes in wind patterns, increased precipitation, and rise in the severity, the intensity and the frequency of storms and SLR (Ministry of the Environment and Food of Denmark & Environmental Protection Agency, 2015). Additionally, Goodsite et al. (2013) also support that the major climate threat for Denmark is SLR. In the same paper, the authors present estimations from the Danish Meteorological Institute (DMI) upon an average sea level rise that could fluctuate from 0.3 m up to 1 m by the year 2100 (Goodsite et al., 2013). The impacts of the changing climate for Denmark can be significant, since a moderate SLR of 0.5 m could almost double the economic damages of one 100-year storm, with an estimated cost of 5 billion euros (Goodsite et al., 2013).

As Jebens and Sorensen (2017) additionally remark that the type of hazard that can significantly alter or disturb the socioeconomic activities of the country are coastal floods, and they also explain that the impacts of climate change are expected to be increased by the shifts in demography, the steep urbanization and the different development projects that are concentrated on the coastal areas. To conclude, the above mentioned facts signify that coastal flooding is a current and future threat Denmark.

#### 1.5 Legislation and initiatives

The disastrous floods of the last two decades that shocked Europe made it clear that the flood reduction measures were insufficient. Therefore, efforts and measures for flood reduction were embarked by the European Union. As a response to the increased flood events and the related damages, the European Union launched the EU Flood Directive (2007/60/EC) in 2007, with a goal of minimizing flood risk and improving flood management through the union-member states (European Environment Agency, 2011; Jebens et al., 2016; Kundzewicz, 2012; Mostert & Junier, 2010). In accordance with the EU Flood Directive, the union's member states had to carry out a preliminary flood risk assessment in order to identify prone areas to flooding (Kundzewicz, 2012; Mostert & Junier, 2010). In addition, the in 2012 Danish government established that every municipality has to design implement municipal adaptation plans, especially for water related hazards (Sorensen & Jebens, 2015)

In the Danish context, the implementation of the EU Flood Directive comprises of the identification of the flood prone areas, followed by hazard, risk and vulnerability analysis, which the aim to include the results in the municipal adaptation plans (Sorensen & Jebens, 2015). The preliminary risk assessment was conducted by the Danish Coastal Authority (DCA) and the Nature Agency in 2011, proposing ten areas prone to coastal and fluvial floods (Jebens et al., 2016). These 10 areas are illustrated in Figure 1.3 with a red point. The initial assessment was followed by a second assessment phase in 2018, the findings of which suggested five additional areas (Kystdirektoratet, 2018b). The new areas at risk of flooding are depicted with a blue point in fFigure 1.3.

Overall, the flood risk assessment in Denmark appointed 14 areas prone to coastal and fluvial flooding, which require special attention in the planning process. For the identification of the areas at risk of flooding the union member states were compelled to follow a holistic framework, which included hazard, vulnerability and risk maps as the basis for appointing the flood risk areas (Jebens & Sorensen, 2017). For the vulnerability assessment, which is of great interest for this research, the criteria that were taken into consideration are categorized as tangible and intangible (Kystdirektoratet, 2018a, 2018b), and are summarised in Table 1.1.



Figure 1.3: Areas at risk of flooding according to the DCA flood risk assessment. The areas with red are the ones identified through the preliminary risk assessment, while the blue ones represent the areas identified by the second assessment (Kystdirektoratet, 2018b)

Table 1.2: Vulnerability indicators, as they were uses in the vulnerability assessment as part of the flood risk assessment by the DCA (Kystdirektoratet, 2018a).

Population density	Sights of cultural heritage
Land use	Critical infrastructure
Infrastructure	Preparedness
Potential polluting enterprises	Local economic activity

#### 1.6 Relevance of study

According to climate projections, with the expected intensification of climate hazards the social inequalities are expected to continue in the future (Agence Européenne de l'Environnement, 2013). Cutter and Emrich (2006) emphasise that the social vulnerability of the population in total is not distributed equally among individuals, groups or and the different areas of a country or a community. The rise in global population that is expected to

continue in the future, the ongoing urbanization and the high population density, especially in coastal areas, will intensify the social inequalities among individuals (Neumann et al., 2015). Even welfare countries, like Denmark, face social problems or challenges that will continue to occur. In combination with the expected effects of climate change as well as the occurrence of more natural hazards, it is expected to have diverse social and economic consequences within the same society, country or community.

Denmark will experience the adverse impacts in the years to come (Goodsite et al., 2013), and as it is shown in Figure 1.4 the vulnerable areas to coastal flooding are expected to increase by the year 2115 (Jebens & Sorensen, 2017). Due to the EU Flood Directive, the Danish Government appointed officially hazard, vulnerability and risk (Jebens et al., 2016; Jebens & Sorensen, 2017), which is important step towards an effective flood management policy and for flood risk reduction. Jebens and Sorensen (2015) also argue about the importance of quantitative social vulnerability assessments for successful flood risk management, in order to promote a people based mitigation approach.



Figure 1.4: Hazard map of potential coastal flooding in Denmark in the years 2015 and 2115 (Jebens & Sorensen, 2017)

However, the indicators that were used for the vulnerability assessment correspond better to the exposure of the area (physical exposure of the area), and not to the social drivers of vulnerability that affect how a hazard and its impacts are distributed among the population. It seems that regardless the acknowledgement of the importance of the social vulnerabilities, the unequal exposure of population to hazards, like coastal flooding, as well as the distribution of vulnerability, are not fully explored neither in the disaster management field and nor in adaptation policies yet (Agence Européenne de l'Environnement, 2013). To add to this, an interview with the chief resilience of Vejle Municipality, a city that established the first resilience strategy in Europe, she denoted that 'Vulnerability is not a topic that we discuss'. This supports the argument that often the attention to vulnerability is neglected in developed countries, since a common perception is that society can cope with (Jebens & Sorensen, 2017). Thus, a way to measure vulnerability based on socio-economic and demographic characteristics of the Danish population could shed light into the topic and provide a holistic understanding of population

vulnerabilities and their drivers, by providing a deeper understanding of the society and its vulnerability (Jebens & Sorensen, 2017)

#### 1.7 Research questions

The main objective of this research is to quantify social vulnerability of the Danish population to flooding, explore the spatial distribution of it and investigate if the identified areas from the Danish flood risk assessments pinpoint the areas not only with the biggest flood exposure but also the areas where the social impacts can be more devastating. The focus of this study will be placed on the local scale vulnerabilities of the Danish population, since floods risk initiatives are implemented in municipal level in Denmark. By constructing a quantitative social vulnerability metric, namely vulnerability index, this research aspires to explore in depth the socioeconomic characteristics that make Danish population more vulnerable to coastal floods and not focusing solely on the surroundings and the physical characteristics that make a place vulnerable to coastal floods. The findings of this research are expected to provide answers to the following research questions:

- Where do the most vulnerable people to coastal flooding live in Denmark?
- What are the spatial patterns between and among the Danish municipalities?

#### 1.8 Thesis Outline

Given the current state of knowledge regarding coastal flooding in Denmark, this research aims to address vulnerability in a more holistic way, by taking into consideration social and economic indicators to identify the spatial clusters of society most likely to suffer if exposed to coastal floods (Chapter 1). Chapter 2 focuses on the research design, the selected theory of science and methods, and the limitations of the research that will affect the results. Furthermore, since vulnerability literature is vague, and the concepts are interdisciplinary vulnerability, risk, and their components will be discussed and defined in Chapter 3. Additionally, Chapter 3 elaborates on vulnerability theories and assessment models, and describes the theoretical framework of this research. Chapter 4 focuses on the indicators selection and the construction of the social vulnerability index. The results of the vulnerability assessment are presented within Chapter 5. Chapter 6 provides an interpretation of the study and offers suggestions for further research. Finally, Chapter 7 summarises the findings of the research, the potential policy implication and the suggestions for furture research.

## 2 Research Design

Research design encompasses the research strategy that is adopted by the researcher in order to find answers to the research question(s). Therefore, for assessing the social vulnerability to coastal floods in Denmark a mixedmethods research design is selected. A mixed-method research design is the combination of quantitative and qualitative methods and data, with major argument that this research strategy provides a more holistic research strategy (Creswell, 2009). The first section of the chapter discusses the theory of science that is adopted. Additionally, the employed methods are presented and the final section of the chapter focuses on the limitations and the delimitations, which affect the research process and the results of the study.

#### 2.1 Theory of science

#### 2.1.1 Constructivism

Constructivism or social constructivism is one of the research paradigms often used in social research. According to Creswell et al. (2013) in social constructivism people interpret the world based on personal experiences and meanings. These subjective meanings are the generated through social, historical and cultural interactions with others, resulting in multiple realities (Creswell, 2013). Hence the ontological stance in this worldview embraces the existence of numerous socially constructed realities. The epistemological belief from a constructivism worldview is that reality is also constructed through social interactions and knowledge can never be truly subjective (Creswell, 2013). In this research paradigm, the researcher's beliefs, meanings and experiences are significant in the process of knowledge creation. Reality and knowledge are "mind-dependent", since they acquire meaning through articulation and social norms that shape both the language and the way of thinking (Ritchie, et al., 2013). Chipangura et al. (2016) further agree that in constructivism reality and knowledge are generated socially and are fluid. Birkmann et al. (2013) argue that "the concept of vulnerability underscores the social construction of risk". They further propose that in order explore the way risk is constructed the different perceptions of vulnerability, hazard and risk should be investigated (Birkmann et al., 2013). Consequently, by adopting a constructivist stance, the researcher has to explore the multiple constructions of truth about a specific phenomenon.

A common critique when adopting a constructivist worldview refers to the subjectivity of the research and the associated results. By adopting an "empathetic neutrality" within this thesis, the researcher aims at building a common theoretical background for the readers in order to make the subjective personal preconceptions and values transparent (Ritchie et al., 2013). This is achieved though the literature review (Chapter 3), where the state of the art in vulnerability field is presented, as well as the relevant conceptual frameworks. In addition, within this chapter, the research design is discussed and the limitations-delimitations are unveiled. In this way, the researcher reveals their approach and aspires to be as objective as possible.

Within the paradigm of social constructivism, ontology and epistemology of disaster risk are defined as subjective (Chipangura et al., 2016). This worldview represents a shift in the disaster reduction and vulnerability field, a shift that "denaturalizes" disasters (Chipangura et al., 2016; Wisner et al., 2004). Wisner et al. (2004) suggest that risk is inseparable from aspects of social life. They further argue that natural hazards can act as an impetus, but the underlying socio-economic conditions create vulnerability and eventually a disaster (Wisner et al., 2004). Cutter et al. (2009) highlight the contribution of O'Keefe et al.'s (1976) paper in the shift from the

nature of the hazard to the social conditions that shape vulnerability. O'Keefe et al. (1976) explain that "Disaster marks the interface between an extreme physical phenomenon and a vulnerable human population... Without people there is no disaster.". This development points out the importance of the "non-natural" drivers of vulnerability and explains the idea of disaster risk and vulnerability as social constructed terms, since they are generated through interactions in social systems.

By adopting a social constructivist approach, it is accepted that vulnerability and risk are social constructs, which are context dependent and interpreted differently based on the selected discipline, context, scale or geographic area of focus. This paradigm interprets disasters as the outcome of social vulnerability of the population and from the 1980's holds a prominent position in the risk disaster field (Chipangura et al., 2016). Therefore, a social constructivist worldview is adopted in order to accomplish answering the research questions and investigating the spatial patterns of social vulnerability in Denmark. The selected conceptual framework, see Chapter 3 for details, supports the view of disasters generated through the interaction of natural and social factors; hence this is the most appropriate paradigm for the purposes of this research. Additional arguments, linking the choice of the paradigm with the conceptual framework exist in the following chapters.

#### 2.2 Methods

This research adopts a mixed methods research design. This approach is selected since quantitative methods are common for making generalizations for a bigger geographic scale, such as a country, while the qualitative methods are the common choice for investigating people's opinion and perceptions of social phenomena (Walliman, 2011a, 2011b). However, mainly quantitative methods will be used for the quantifying social vulnerability to coastal flooding, since the vulnerability metric that is selected supports the selection of quantitative data. The qualitative methods will be used to explore perceptions of vulnerability for the problem formulation, the selection of data and the adaption of vulnerability indicators based on relative literature and documents. In general, a social constructivism worldview uses qualitative methods to explore individual's perception and understandings for a specific topic. However, the selected theory of science is related directly to the understanding of vulnerability and risk and the selection of the vulnerability framework and indirectly with the selected methods. It is well acknowledged that the use of mainly qualitative methods could help answer the research questions, but since this research will explore the vulnerability in national level, it was decided to stick with one type of methods as the main one.

#### 2.2.1 Qualitative Methods

#### 2.2.1.1 Literature Review

Fink (2005) defines literature reviews as "a systematic, explicit, and reproducible method for identifying, evaluating, and synthesizing the existing body of completed and recorded work produced by researchers, scholars, and practitioners.". Literature review is an important method for summarising the current knowledge and identifying gaps in the existing literature, as part of the initial research steps.

Therefore, literature review was conducted for the purposes of this thesis, and serves as the backbone of the theoretical background. Firstly, the literature review is a way to summarise the state of the art in vulnerability field, present the findings to the readers, while it makes possible to build a basic understanding around the complex issue of disaster risk and vulnerability. Therefore, is an integral step for building the theoretical

background of the research, presenting readers with the same information as the researcher, and clarifying the position of the researcher. Moreover, by reviewing the literature is possible to define the key concepts around the topic of vulnerability, and creates a common terminology for the investigated term among the researcher and the readers.

The literature review comprises chapter 3, which focuses on presenting the work of the pioneers of the field, justifying the selection of specific definitions and associated theories and framing research arguments, instead of conducting an extensive review of a long literature list. In order to conduct the literature review, a systematic search of keywords in the search engine of Aalborg University, Google Scholar and other databases was implemented. The literature that was selected was peer reviewed. Finally after detailed reading of the selected sources, the literature review chapter was synthesized, and is presented in the next chapter.

#### 2.2.1.2 Semi-structured interview

According to Clifford et al. (2010) an interview can be defined as "verbal interchange where one person, the interviewer, attempts to elicit information from another person". A semi-structured interview has been conducted for this research to provide useful insights regarding the importance of vulnerability assessment and identification of vulnerability population on policy level in Denmark.

Semi-structured interviews provide bigger flexibility than structured interviews, while at the same time have a defined structured as a basis. This method allows the researcher to discuss topics of interest that derive from the interviewee during the interview process. As Bernard (2006; 2006) suggests semi-structured interviews are suggested to gain knowledge in a one-time meetings with the interviewee, due to the great degree of freedom and flexibility. The overall aim with the use of the interview is to gain a better understanding of the current interpretation of vulnerability as part of climate adaptation policies in municipal level in Denmark. For the semi-structured interview an expert in the field of climate adaptation was selected, while the findings of the interview were used as inputs in the formulation of the research problem.

#### 2.2.1.3 Document Analysis

Document analysis is a qualitative method that is been used for interpreting the meaning of documents regarding a topic of interest (Bowen, 2009). Secondary document data are analysed with this method. The data include public records from the Danish Government, the DCA, the EU and the Storm Council. With the use of this method, documents related to vulnerability and flood risk reduction were reviewed as part of the process for the selection of the indicators to quantify social vulnerability to coastal floods in Denmark.

#### 2.2.2 Quantitative Methods

#### 2.2.3 GIS

Quantifying social vulnerability on a national level can be complex and is a process that requires large datasets. Thus, Geographic Information Systems (GIS) are used for utilizing and analysing mutli-scalar data. The GIS serve as a tool for the implementation of the social vulnerability model in Denmark, and permit the visualisation of the results of the model in a comprehensive way for practitioners and planners. Moreover, with the use of overlay analysis, layers that correspond to the social drivers of vulnerability can be explored in the scale of

choice, making it possible to monitor the spatial patterns, and identify the "hotspots" of the vulnerability that require attention in the planning process. Therefore, GIS are often used in vulnerability assessment cases.

For the purposes of this research all GIS processing was conducted within ArcMap 10.6. The data that were used are presented in Chapter 4, along with the detailed data analysis. The GIS were selected for the analysis part of the research in offer to present the vulnerability scores and allow comparison between municipalities. After the calculation of the vulnerability scores for each municipality, the following step was to import the data into the GIS in order to allow translate vulnerability scores into spatial information.

#### 2.2.4 Principal Component Analysis (PCA)

Principal component analysis is a very useful method in order to manipulate large set of variables. PCA is a statistical method for reducing a large set of correlated variables into a smaller number of variables, by identifying the uncorrelated values that can represent the majority of the information of the original set of variables (Dunteman, 1989). The derived variables are named principal components (Dunning & Durden, 2011). The goal of this method is to create a smaller set of variables for facilitating the analysis process.

The method is used for identifying the most important variables, which can explain the drivers of social vulnerability. These variables correspond to the indicators used in the calculation of the vulnerability to coastal floods for each municipality in Denmark.

#### 2.3 Limitations

Every type of research and its results are the outcome of multiple decisions made by the researchers. These decisions might include potential limits associated with the selected methods, called limitations. Another type of decisions that have a crucial impact on the implementation of the research are decisions around the research topic, these are called delimitations. Once those decisions are acknowledged and communicated to the readers, it is possible for the researcher to obtain a more neutral position. By clarifying the decisions that have been made before or during the research, it is possible for the reader to understand the choices that have shaped the research findings. Moreover, these decisions can be seen as suggestions for future research around the topic, since different research design and methods imply different findings. Overall, by identifying the limitations and the delimitations of the research, the scope of the research and how a problem is viewed and investigated it becomes clearer for the reader and it is important precondition for producing valued knowledge.

#### 2.3.1 Limitations

- Qualitative data are more difficult to produce generalized conclusions, since the interpretation of them includes subjectivity;
- Quantitative data cannot easily shed light on individual's perceptions, experiences and opinions;
- Data availability and scale of the available data can disturb the social vulnerability assessment and the SoVI scores. Scale of data can vary from local, regional or national level for example;
- SoVI, like all the vulnerability metrics, is a relative measure of vulnerability.

#### 2.3.2 Delimitations

• The selection of the interviewee for the semi-structured interview;

- Researcher subjectivity is significant in the selection of the theory of science, methodology and the vulnerability conceptual framework ;
- Adaptation of the indicators and of the hazard-of-place model to fit the Danish context;
- The use of mainly quantitative data;
- Language barrier in the analysis of the documents from Danish sources;
- Indicators selected to fit the context of coastal flooding in Denmark might be inappropriate to measure social vulnerability floods for another country;
- In this thesis only the social vulnerability aspect of the conceptual model is examined.

## **3** Theoretical framework – Key Concepts

The negative impacts of coastal flooding, like other types of natural hazards, vary according to the physical exposure of the area as well as the vulnerability of the people. Vulnerability is distributed unequally among individuals or groups of an area based on specific characteristics that they share, which affect their vulnerability level. Vulnerability is context dependent and therefore multiple definitions exist based on the discipline and the topic of research (Füssel, 2007). The way a researcher interprets vulnerability and risk affects significantly the way vulnerability is conceptualised and therefore quantified. Rygel et al. (2006) state that the first steps in vulnerability assessment is to select a conceptual framework and then define the associated terms. Thus, it is crucial to define the terms properly and create a clear understanding of what they stand for in this research, before proceeding in the conceptual framework selection. This chapter reviews the development of the concepts of risk, vulnerability, exposure and susceptibility. The following sections are advised to be seen in accordance with the Appendix I, II, III and IV for detailed presentations of risk, vulnerability, exposure and susceptibility definitions respectively.

#### 3.1 Risk

Risk in its simplest form is defined as a measure of the threat posed by a hazard in a specific area (Cutter, 1996). The most traditional definition of risk was proposed by United Nations (UN) (1993), where it is defined as "Risk = Hazard x Vulnerability". A hazard refers to a natural, a technological or a human made event, and has the potential to disturb the social and economic situation of people in a certain area (Fitton, et al., 2018; Villagran de Leon, 2006; Wisner et al., 2004). The same interpretation of risk was selected again by the UN after eleven years, when they launched the global risk reduction initiative (2004). In this sense, risk has two components; hazard and vulnerability, which constitutes the interpretation commonly used in the risk and disaster reduction field, both in academia and on policy level. This definition implies that risk is defined by the magnitude or frequency of a hazard occurrence and the characteristics of the people that make them more vulnerable. This definition was prevalent in the disaster risk field until the mid-1980's and early 1990's (Cardona, 2003; Chipangura et al., 2016). Chipangura et al. (2016) state that from a realist worldview risk is perceived as an "objectively identifiable phenomenon", while Wisner et al. (2004) advocate that from this perspective risk is objective and independent of social and cultural conditions. To add to this, Füssel (2007) characterises this definition as "geocentric", because the focus is placed on the nature's role as the driving force of a disaster, neglecting the contribution of the human actors.

From a different perspective, Villagran de Leon (2006) interprets risk as the outcome of hazard occurrence, vulnerability of people and insufficient preparedness on behalf of the community. With this definition preexisting conditions are considered to affect how the hazard is anticipated and the distribution of the hazard impacts in a community. Blaikie et al. (1994) and Dilley et al. (2005) suggest exposure as an additional component of risk, while Davidson et al. (1997) and Bollin et al. (2003) add adaptive capacity as a component in the risk definition. Despite the contradictory components that these authors suggest, the proposed definitions share a common feature. These definitions propose that human actions affect the generation of risk, adopting a more constructivist approach. Hence, risk is influenced by social conditions and interactions.

Despite the differences among the definitions and the adopted worldview, a point of convergence is the probabilistic nature of risk. Alexander (2000) argues that risk is "the likelihood, or more formally the probability, that a particular level of loss will be sustained by a given series of elements as a result of a given level of hazard". The UNISDR (2004) supports that risk is the probability of the unwanted impacts that are generated by the interaction of hazard and vulnerability. While, Downing et al. (2001) refer to risk as the "probability of hazard occurrence". However, the interpretation of the term "probability" in the above mentioned definitions can vary and therefore can create confusion, since probability refers both to the occurrence of the hazard and to the potential impacts of it.

Within this research, the term risk will refer to the interaction of hazard, exposure and vulnerability, according to the definition proposed by Winser et al. (2004). This definition highlights the contribution of the social, economic and cultural drivers of vulnerability as important factors in the construction of risk and aligns best with the theoretical worldview that is adopted for this research.

#### 3.2 Vulnerability

In general, vulnerability is defined as the predisposition of an individual, or a group, to be harmed due to different risks, hazards or stressors (Cardona, 2003; Kaperson & Kasperson, 2001). Terms often associated with vulnerability are susceptibility, sensitivity, exposure, adaptive capacity and resilience, depending on the context of application each time. Despite the abundance of literature on vulnerability and its components, the only consensus that has achieved is about the fuzziness and the broadness of the definition (Adger et al., 2004; Birkmann et al., 2013; Costa & Kropp, 2013; Joakim, 2017; O'Brien et al., 2004). The confusion around vulnerability understanding and vulnerability assessment methods continues to grow not only between researchers of different disciplines, but also between practitioners and policymakers around the word (Adger et al., 2004; Costa & Kropp, 2013). The ambiguity of the term vulnerability can be attributed to the fact that many groups utilize the same concept giving a unique context-specific meaning, mainly because of the need to describe a unique situation of hazard and risk (Villagran de Leon, 2006). Therefore, many studies point out the need for a holistic approach for vulnerability (Birkmann et al., 2013; Cardona, 2003; Birkmann, 2006). For detailed reviews of vulnerability conceptual frameworks please advise the work of Birkmann (2006b), Füssel & Klein (2006), Villagran de Leon (2006) and Cutter et al. (2009). In Appendix III, 24 definitions of vulnerability are gathered, reviewing vulnerability research from 1981 until 2016.

Vulnerability concept originates from the disaster management research, which is considered to have started in the decade of 1940 (Cutter et al., 2009). White and Haas (1975) investigate the connection between people living in hazard areas and the vulnerability, which resulted in the generation of risk-hazard paradigm. This theoretical

approach is also observed in the work of Burton et al. (1978). By adopting a nature oriented approach, the hazards posed by nature as considered the main threat and the driver of vulnerability (Eakin & Luers, 2006), hence this approach is often described as nature centred (Cutter et al., 2009). Birkmann (2005; 2006b) argues that initially the understanding of vulnerability was limited, since it was perceived as a component of risk (Figure 3.1., first shpere) Scholars and policy makers in the Climate Change Adaptation (CCA) field of research are proponents of the risk-hazard paradigm (Breil et al., 2018; Cardona, 2003), where vulnerability is the outcome of exposure, sensitivity and adaptive capacity (Birkmann et al., 2013). This theoretical approach towards risk and vulnerability is illustrated also in the work of UN (1993) and the IPCC (2007; 2001a).

O'Keefe et al. (1976) argue to the natural focus of the risk-hazard paradigm by proposing that "disaster marks the interface between an extreme physical phenomenon and a vulnerable human population. It is of paramount importance to recognize both of these elements. Without people there is no disaster.". O'Keefe et al. (1976) first adopt the political economy school of thought. Overall, the political economy approach investigates the social, economic and cultural aspects of life in order to interpret vulnerability distribution and changes among different populations and the capacity of individuals to anticipate the impacts of a hazard (Cutter et al., 2009; Eakin & Luers, 2006). Additionally, the political economy approach connects vulnerability with shifts in the political landscape and the distribution of power (Birkmann et al., 2013). Hewitt's (1983) work criticise the overlay focus on the natural side of disasters, proposing that hazards are not natural. Chambers (1989), Watts and Bohle (1993), Blaikie et al. (1994) and Wisner et al. (2004) adopt also the political economy approach in their conceptualization of vulnerability and risk (Figure 3.1, third sphere). While, Wisner et al. (2004) propose that vulnerability comprises the probability of human lives to affected by a hazard, introducing in this way a definition that takes into consideration human actions, see the second sphere in Figure 3.1. Füssel (2007) further adds that the structure of society creates the preconditions for generating vulnerability, while accessibility to resources or information, political instability or conflicts affect people's lives and thus shape their vulnerability. Therefore, Cutter et al. (2009) characterise this approach as "human-centric", since human capacity and interactions are important components in this vulnerability approach.

The next theoretical development in the field of vulnerability research is dominated by a hybrid approach. Cutter (1996) propose the hazard of place model as a place-based approach, combining components from the two previous theoretical approaches. This integrated or holistic approach of dealing with vulnerability points out the dynamic nature of vulnerability and the importance of potential feedback loops in the vulnerability assessment (Cutter, 1996). According to the hazard of place model vulnerability is the result of the interaction between social and biophysical factors, while it suggests that vulnerability is dynamic (Cutter et al., 2009). An additional development in the vulnerability understanding is illustrated in the fifth sphere of Birkmann (Figure 3.1); where vulnerability comprises of additional components, hence the name multiple structures. Turner et al. (2003) define vulnerability as the outcome of the interaction of; exposure, adaptive capacity, susceptibility and coping capacity. The last sphere represents the final widening phase of vulnerability concept (Figure 3.1). As Birkmann (2005; 2006a) explains, researchers like Cardona (2003; 2012) and O'Brien et al. (2004) adopt a "multidimensional" vulnerability perception; a way of understanding that implies a focus besides the physical dimension.



Figure 3.1 The widening of vulnerability concept, adapted from Birkmann (2005b; 2006). According to the figure, vulnerability has evolved from component of risk to a concept with multiple dimensions.

Despite, the development in the field of disaster management and vulnerability that brought attention to the social factors that affect vulnerability generation, researchers continue to differentiate biophysical and social vulnerability. Brooks (2003) remark that biophysical vulnerability comprises of the hazard event, exposure and sensitivity of a system. Brooks (2003) further point out that biophysical vulnerability implies the physical aspect of the hazard impacts, while Holand et al. (2011) claim that the term is closed related with the risk definition, as a function of hazard and vulnerability. On the contrary, social vulnerability is defined as "[T]he characteristics of a person or group and their situation that influence their capacity to anticipate, cope with, resist and recover from the impact of a natural hazard (an extreme natural event or process)" (Wisner et al., 2004). Cutter and Finch (2008) define social vulnerability as "a measure of both the sensitivity of a population to natural hazards and its ability to respond to and recover from the impacts of hazards". To add to this, Cutter et al. (2003) advocate that social vulnerability is influenced also by the social inequalities that are inherent in social systems. Therefore, one could argue that social vulnerability is a characteristic that can be found within social systems, since it is embedded in the social structure of the systems. In the context of this research, when using the term vulnerability it will refers to social vulnerability. As Myers et al. (2008) conclude that this notion of vulnerability is supporting the social construction of disasters though the social conditions that make individuals more or less vulnerable, while disasters are not considered natural phenomena and the human actors are not perceived as incapable to act, respond or adapt.

#### 3.2.1 Vulnerability models

Besides the theoretical developments in vulnerability conceptualization, in order to accomplish assessing vulnerability a conceptual model is required. With acknowledgement of the existence of multiple vulnerability models, the two most cited models are presented. They are selected due to their longevity and their applicability by many scholars. The conceptual models of vulnerability aim at measuring vulnerability, while facilitating the indicator selection process (Birkmann, 2006a). Since, it is already stated that this research adopt a social constructivist view towards disasters and vulnerability, the presented models correspond to this perception of

vulnerability. The first section is dedicated to Blaikie's et al. (1994) "Pressure and Release" model, following by the hazard of place model, developed by Cutter (1996). Also, this section provides the arguments for the selection of the hazard of place model.

#### 3.2.1.1 Pressure and release model

Pressure and release (PAR) model is a well acknowledged model among the plethora of the existing conceptual frameworks. This model was introduced by Blaikie et al. (1994), and developed further by Wisner et al. (2004). PAR model is built upon the premise that disasters are generated through the interaction between the exposure to a hazard and multiple socio-economic factors that shape human vulnerability (Singh, 2014; Wisner et al., 2004). Wisner et al. (2004) state that disasters are the outcome of the interaction between extreme natural events, named hazards, and vulnerability that is generated through social processes. Wisner et al. (2004) further highlight that "disasters are a result of the interaction of both; there cannot be a disaster if there are hazards but vulnerability is (theoretically) nil, or if there is a vulnerable population but no hazard event.". Therefore, this model focuses on the development of vulnerability, which is generated by hierarchical casual factors, which are the root causes, the dynamic pressures and the unsafe conditions (Figure 3.2).

Root causes of vulnerability constitute the large scale political, economic and social processes that influence the distribution of different types of resources among the members of a social system, and constitute the global social fabric (Singh, 2014; Wisner et al., 2004). Dynamic pressures are "processes and activities that 'translate' the effects of root causes both temporally and spatially into unsafe conditions", while unsafe conditions are defined as "the specific forms in which the vulnerability of a population is expressed in time and space in conjunction with a hazard" (Wisner et al., 2004). Blaikie et al. (1994) compliment the PAR model with the development of the Access Model. which is further developed in the work of Wisner et al. (2004). Access model "deals with the amount of 'access' that people have to the capabilities, assets and livelihood opportunities that will enable them (or not) to reduce their vulnerability and avoid disaster" (Wisner et al., 2004), providing a comprehensive way to link socio- environmental processes with a disaster event.



Figure 3.2: Pressure and Release Model (Wisner et al., 2004)

PAR model is mainly used to compare the vulnerability of different social groups, since it has the ability to identify the causes of vulnerability in a comprehensive way (Turner et al., 2003). Furthermore, PAR accomplishes highlighting the societal conditions that lead to the manifestation of unsafe conditions and generation of vulnerability, but fails at dealing with the biophysical side of vulnerability (Turner et al., 2003). Another limitation of the model is that vulnerability is perceived as static (Marchand, 2009). Consequently, this model is not suitable for quantifying vulnerability to coastal floods in Denmark, since it is contradictory to the adopted notion of vulnerability as dynamic.

#### 3.2.2 Hazard of place model

Within this research, the interpretation of vulnerability and risk derives from the hazard-of-place model (Cutter, 1996). This conceptual framework was developed by Cutter (1996) and links social and biophysical vulnerability together in order to produce the overall or place vulnerability (Figure 3.3). The model was developed further by Cutter et al. (2000; 2003) providing a holistic approach towards disaster management. Cutter et al. (2000) argue that "the interplay of social, political, and economic factors - interacting separately, in combination with one another, and with the physical environment -creates a mosaic of risks and hazards that affect people and the places they inhabit.".

The model illustrates that risk, which is defined as "the likelihood of the event occurring and includes three sub elements: the potential sources of risk, the impact of risk itself, and an estimate of the frequency of occurrence.", interacts with possible mitigation actions and combined together they define the hazard potential (Figure 3.3). Furthermore, hazard potential of the system has a geographic component, for instance proximity to a hazard or other physical characteristics, and a social one, named the social fabric. Social fabric in this framework represents experiences, perceptions, and social characteristics that affect in a great proportion how hazard can distribute. Both components generate later the biophysical and the social vulnerability respectively. These scores combined constitute the total vulnerability or place vulnerability. Finally, place vulnerability is connected with feedback loops with risk and mitigation, which represent the dynamic nature of the system.

The hazard of place model is selected as the conceptual framework for the purposes of this research. This decision was made upon the fact that this model has a more "geographical" approach than other conceptual frameworks (Joakim, 2017). The model places geographic scale as a major component of the vulnerability assessment (Cutter et al., 2000). To add to this, Cutter (1996) further points out that vulnerability is quantified and interpreted in a specific geographic scale, thus this methodology takes into consideration both physical and social characteristics of a place and its people for assessing vulnerability. Furthermore, the hazard of place model is a dynamic and recognizes the feedback loops among the components of risk and hazard, while it acknowledges people as active actors of the system (Cutter et al., 2000; Cutter et al., 2003; Joakim, 2017). Therefore, the basic premise of the hazard of place model corresponds both to the understanding of vulnerability as a social construct and the inherent dynamic nature of the term. Finally, the model with a variety of vulnerability indicators support the use of mixed methods in quantification of social vulnerability. This research focuses on the social vulnerability side of the model (Figure 3.3); therefore, the place vulnerability for Danish municipalities cannot be estimated without further research.



Figure 3.3: The hazard-of-place model (Cutter et al., 2000; Cutter, 1996; Cutter et al., 2003). This research aims at investigating the "social fabric" in Denmark in order to accomplish measuring social vulnerability.

#### 3.2.2.1 Social Vulnerability Index (SoVI)

In order to be able to measure social vulnerability Cutter et al (2003) developed a quantitative comparative metric; the Social Vulnerability Index (SoVI). With the use of socio demographic data obtained from the census of United States, Cuter et al. (2003) manage to identify 11 indicators, which conceptualise the drivers of vulnerability in the specifc context. SoVI generates relative social vulnerability scores of an area with the use factor analysis of the demographic factors.

Dunning and Durden (2013) suggest that a significant advantage of this metric is that it allows to elaborate further on the drivers of vulnerability, instead focusing solely on the overall score. Moreover, the identification of the drivers, or the main indicators that affect social vulnerability, permits the comparisons not only of the total scores, but also the driving forces of vulnerability in a specific context. Kuhlicke et al. (2011) argue that SoVI is one of the vulnerability indices that can be transferable to different types of hazard and geographic contexts as well. Therefore, studies employing the SoVI as the selected vulnerability metric and implemented in Norway (Holand et al., 2011), Brazil (Hummell et al., 2016), China (Chen et al., 2013) and Portugal (Guillard-Goncąlves et al., 2015), point out the transferability of the metric to other geographic and social contexts.

However, the statistical data process and the requirement of large data sets can be a daunting task both for researchers, but mainly for hazard and disaster reduction managers (Evans et al., 2014). Hence, Cutter et al. (2011) created a simplified metric based on the SoVI metric, the SoVI Lite (Cutter, et al., 2011). SoVI Lite utilize a smaller data set and the calculation of vulnerability is measured without statistical analysis (Cutter et al., 2011; Dunning & Durden, 2011; Evans et al., 2014). Overall, SoVI Lite comprises a version of the initial SoVI aimed a reducing the complexity of the statistical analysis procedure (Evans et al., 2014). This research will be conducted with the use of the SoVI Lite metric, which will be customized in terms of factors and indicators to be tailored to the Danish context. Detailed description of the customization of the metric, as well as the data collection and analysis, are discussed in the following chapter.

#### 3.3 Exposure

The definitions that stand out in disaster risk literature define exposure as a way to measure a hazard (Birkmann et al., 2013; Blaikie et al., 1994). While, in the climate change community exposure is interpreted as the degree to which climate change can have adverse impacts on humans lives (IPCC, 2007; IPCC, 2001a). In another viewpoint, exposure is the "external side of vulnerability" (Bohle, 2001; Watts & Bohle, 1993). From this perspective, exposure denotes the negative impacts of a hazard that potential can alter the social conditions of a system. Furthermore, Turner et al. (2003) propose that exposure is the outcome of the interaction between inherent characteristics of the social system and the hazard occurrence. Costa and Kropp (2013) argue that there is an overlap in the use of term. In the disaster risk context exposure can denote both the external side of vulnerability and the internal side of risk. In this research, from now on, the term exposure will refer to "a component of vulnerability addresses not only the extent to which a system is subjected to perturbation, but also the degree and duration of that perturbation" (Adger, 2006).

Social vulnerability focuses on the social conditions that reduce the ability of individuals to anticipate the hazard event, instead of the importance of the negative consequences of a hazard (Costa & Kropp, 2013). Therefore, exposure, from this viewpoint, is an internal characteristic of the social system itself. For the purposes of this research, the authors adopt the definition proposed by Birkmann et al. (2013), where exposure is defined as "the extent to which a unit of assessment falls within the geographical range of a hazard event. Exposure extends to fixed physical attributes of social systems (infrastructure) but also human systems (livelihoods, economies, cultures) that are spatially bound to specific resources and practices that may also be exposed".

#### 3.4 Susceptibility

In the disaster risk literature, susceptibility can be found as a component of risk in the definitions of White et al. (2005) and Scheuer et al. (2011). In this view, susceptibility represents the internal side of vulnerability, and refers to "the social, economic, political, psychological and environmental variables that intervene in producing different impacts amongst people with similar levels of exposure" (Costa & Kropp, 2013; P. White et al., 2005). Moreover, Cutter and Emrich (2006) view susceptibility as a component of vulnerability. On the other hand, the climate change community uses the term sensitivity for the same purposes (Costa & Kropp, 2013). Sensitivity, therefore, represents the internal side of vulnerability in accordance with the definitions of Turner et al. (2003) and IPCC (2007). Turner et al. (2003) suggest that sensitivity is affected significantly by the human-environmental interactions that take place in an exposed system. IPCC assigns sensitivity to the degree that a system is prone to climate change impacts.

Even though there is no universal definition among the different disciplines, authors seem to agree in the nature of susceptibility characteristics, since Birkmann et al. (2013), White et al. (2005) and Fekete (2010) agree that both social, economic and physical characteristics contribute in a system's susceptibility. Costa and Kropp (2013) remark additionally that susceptibility and sensitivity meaning coincide among the multiple definitions. In this research, the term susceptibility will correspond to the definition of Birkmann et al. (2013), who interpret the term as "the predisposition of elements at risk (social and ecological) to suffer harm".

#### 3.5 Glossary

Chapter 3 has discussed the evolution of the literature in the field of risk and vulnerability. The main points of the chapter are that describe the working approach are summarised as follows:

• Within this research risk will refer to "Risk as consisting of two components. The first component is a measure of the natural hazard. The second, vulnerability itself, is equivalent to capacity and is, they argue, largely determined by socio-economic structure and property relations", as proposed by Blaikie et al. (1994);

• Vulnerability corresponds to social vulnerability, and is defined as "a measure of both the sensitivity of a population to natural hazards and its ability to respond to and recover from the impacts of hazards", as proposed by Cutter and Finch (2008);

• The working approach within this research is based on the hazard of place model, because it is a place based approach that allow comparisons among places, while combines biophysical and social factors;

• Exposure corresponds to biophysical vulnerability, as it described by the hazard of place model. This research focuses on the social vulnerability to coastal floods; therefore exposure is not measured;

• Susceptibility stands for "the socio-economic and environmental characteristics of a system that shape the degree of impacts", according to Fekete (2010).

## 4 Developing the Social Vulnerability Index

As it is stated in the previous chapter, this research project will conduct the assessment of the social vulnerability to coastal floods in Denmark by adopting the hazard-of-place conceptual model, which will be operationalised with the use of the SoVI Lite metric. Since SoVI is designed to use census data from the United States and is conformed to this specific socioeconomic context, the application of the metric requires an adaptation in terms of the indicators. Therefore, the indicators used in SoVI metric will be adapted based on the available data and local characteristics that affect how vulnerability is created in Denmark. The focus of this chapter is placed on the steps taken to develop an index of social vulnerability to coastal floods in Denmark. The first section presents the indicators utilised within the SoVI Lite and the rationale behind their selection, and discusses how the indicators are relevant to the Danish context. Following this, the data collection and the data sources are detailed.

#### 4.1 Social Vulnerability Indicators

Cutter et al. (2008) define vulnerability indicators as "Quantitative measures intended to represent a characteristic or a parameter of a system of an interest". Cutter et al. (2009) further add to the definition the use of a single value for each indicator. Burton et al. (2018) argue that indicators cannot capture and quantify the overall vulnerability of a population, instead they should be seen as a way to understand at an extent the factors that construct vulnerability in this case. Hence, a meticulous task for the purposes of this research is to select indicators that will explain the social fabric of the Danish population.

The social vulnerability assessment of the coastal floods in Denmark is based on the SoVI Lite (Cutter et al., 2011). Subsequently, the SoVI Lite indicators will be used as a guideline for the indicators selection. The identified indicators by Cutter et al. (2011) are presented in Table 4.1. Along with the indicators, the rationale behind the selection their selection is presented, with the support of relevant literature. The rationale is mainly based on the detailed work of Cutter et al. (2003) with the addition of recent findings of supporting literature.

Indicators	Rationale of selection	Supporting Literature
Population in poverty (positive)	Economic status of an individual, or a community, influences heavily the ability to respond and recover from a natural hazard. Wealthy people have the economic resources to anticipate the impacts of a hazard, due to savings, insurance of their material assets, while poor people are more prone to suffer and recover after a hazard. Moreover, poor people usually dwell in older houses, increasing the total cost of material damages.	(Blaikie et al., 1994; Burton, et al., 1993; Clark et al., 1998; Cutter et al., 2000; Cutter et al., 2003; Fothergill & Peek, 2004; Hewitt, 1997; Morrow & Peacock, 1997; Morrow, 1999; Platt, 1999; Rygel et al., 2006; Fitton, et al., 2018)
Population over 65 (positive)	Elderly population are prone to harm during or after a hazard, due to problems associated with health, restricted mobility, unwillingness for evacuation due to higher attachment to their property and have lower ability to respond and recover. Early	(Clark et al., 1998; Cutter et al., 2000; Cutter et al., 2003; Fitton, et al., 2018; Hewitt, 1997; Morrow,

Table 4.1: Vulnerability indicators used by Cutter (2011) that have relevance to the Danish context. The brackets in the first column stipulate the relationship between vulnerability and the indicators. A positive relationship indicates that the indicator increase vulnerability, while a negative relationship decreases vulnerability.

	warnings might be insufficient for them, due to their non familiarity with technology.	1999; O'Brien & Mileti, 1992; Peacock, et al., 1997; Rygel et al., 2006)
Urban population (positive)	The high population or dwelling density might create difficulties in the evacuation process in case of emergency. Moreover, areas with lower dwelling density might be unprotected with defensive measures, like dikes in the case of coastal floods. To add this, rural dwellers often are employed in the primary sector; hence, they are often dependent on the local resources.	(Cova & Church, 1997; Cutter et al., 2000; Cutter et al., 2003; Fitton et al., 2018; Puente, 1999)
Dwellers in rental houses (positive)	Usually renters have lower income than house owners. They dwell in rental properties either due to lack of the resource to buy or because there are transients. Often renters are perceived to have higher vulnerability, due to lower economic power and lack of access to recovery resources, and limited social networks compared to house owners.	(Cutter et al., 2000; Cutter et al., 2003; Fitton et al., 2018; Heinz & Tufford, 2001; Morrow, 1999)
Female Population (positive)	Gender is an important factor in the distribution of vulnerability. Women have higher vulnerability than men, especially because of lower average income, employment in sector-specific positions and their family responsibilities. In a cause of evacuation during or after a disaster, women tend to prone to harm, since they are the one that have to supervise and help the vulnerable family members, like kids and elderly.	(Bianchi & Spain, 1996; Blaikie et al., 1994; Cutter, 1996; Enarson & Morrow, 1997; Enarson & Scanlon, 2002; Fothergill, 1996; Hewitt, 1997; Morrow, 1999; Peacock et al., 1997; Rygel et al., 2006; Fitton, et al., 2018)
Hispanic population (positive) Native American population (positive)	Racial and ethnic minorities often experience discrimination and marginalization. These people tend to have lower income and education level, higher unemployment rates, and live in areas with lower rent, resulting in clustering of minorities and weaker social networks. Moreover, many immigrants do not speak the language of the host country, increasing their vulnerability during or after a hazard.	(Bianchi & Spain, 1996; Clark et al., 1998; Cutter et al., 2003; Fothergill, 1996; Fothergill & Peek, 2004; Peacock et al., 1997; Rygel et al., 2006; Yelvington, 1997)
Population in nursing facilities (positive)	People in nursing facilities are vulnerable to hazards. Due to physical or mental disabilities, their response to disasters can be limited. In the evacuation process, these individuals need to be prioritized, due to potential problems in their mobility and they also need assistance for their recovery. Their recovery is also heavily dependent upon the social services; hence a hazard can interfere with their wellbeing and their recovery.	(Cutter et al., 2003; Morrow & Phillips, 1999; Rygel et al., 2006)

The indicators presented in Table 4.1 highlight that socially and economically marginalized populations are more vulnerable during or after a natural hazard (Cutter et al., 2003; Fitton et al., 2018). However, as it was discussed above, some of these indicators do not correspond in the Danish societal context. Hence, some indicators need to be adjusted in order to accomplish representing the marginalized Danish population. These indicators are the following:

- Hispanic population: In general, it is agreed that Denmark is characterised by a homogenous population with strong cultural identity (Moore, 2010; Wren, 2001). Wren (2001) claims that Danes are traditionally tolerant people with liberal ideas, however in the 1980's a shift in society changed the social equilibrium. In the recent years, many arguments have been raised around the integration of immigrants, racism and social marginalization. Wren (2001) argues about the rise of "cultural racism" in Danish society. Moore (2010) discusses about the xenophobia in Denmark and the new public discourses that have emerged and illustrate the shift in the mentality towards immigrants and refugees. Even though there are multiple ethnic groups in Denmark, the new discourses can be characterized as anti-Muslim (Wren, 2001). Muslims in Denmark are represented by different ethnic groups, which are generalized in the literature as Non-Western immigrants. Moreover, areas with high concentration of Muslims or Non-Western population, comprising over than 50% of the population, if combined with high criminality rates, unemployment, or low educational levels are identified as ghettos by the Danish Government (Boligministeriet, 2017). Therefore, it can be argued that Non-Western population is the most vulnerable racial minority in Denmark, which corresponds to the use of Hispanic population in SoVI Lite methodology.
- *Native American population*: The SoVI Lite metric includes Hispanic and Native American population as vulnerability indicators, underlying the diverse ethnic background of the American population. Discrimination, social isolation, lack of knowledge of the local language can increase the vulnerability of ethnic groups, as Table 4.1 shows. However, these indicators have to be adapted in order to reflect the societal context of the Danish population. Firstly, Greenlandic and Faroese people who live in Denmark represent the "Native American" indicator. One could argue that these two ethnic should be taken into consideration with regard the home-nation's relationship with Denmark. Both Faroese and Greenlandic people are considered Danish citizens; however, there are cultural and linguistic differences among them and the Danes.

The indicators that Cutter et al. (2011; 2003) identify illustrate the factors that influence vulnerability to a range of different natural hazards. Consequently, coastal flooding vulnerability studies tailor the indicators to the specific hazard type. Based on literature findings, an important factor in the case of coastal floods is car ownership. Singh et al. (2014) and Dwyer et al. (2004) point out that car ownership is among the factors that has an influence on human vulnerability. Hurricane Katrina showcases that access to a car is a key factor in generating social vulnerability (Rufat et al., 2015). Moreover, Fitton et al. (2018) advocate that that access to a car can decrease significantly the vulnerability reduces local dependency on services or resources (Fitton et al., 2018). Therefore, in a case of coastal flooding, car ownership is important and might be helpful in the recovery process after a hazard. For instance, if an individual need to be hospitalized in a distant facility, car can reduce distance and time. Another example refers to local goods and groceries, which after a disaster can be limited and more expensive. In this case, car ownership reduces the cost and also increases the choices for an individual. As a result, car availability is considered to decrease social vulnerability to coastal floods and is selected to be included in the vulnerability index.

Another indicator that has been omitted from the SoVI Lite metric (Cutter et al., 2011; Evans et al., 2014) is occupation. According to Cutter et al. (2003) employment based on resource extraction is vulnerable to natural

hazards. Especially in the case of coastal flooding, fisheries, agriculture and forestry might have greater difficulties recovering from a flood hazard, when compared to service oriented positions (Cutter et al., 2003). In Denmark, only 3% of the land is built up, while the rest is used for agricultural production and other locally based economic activities, including fisheries, forestry and farming (Bruun et al., 2012). Hence, one could argue that many people are employed in these industries, with occupation affecting significantly individual's capacity to recover from a coastal flood. People employed in these sectors are more prone to lose their jobs and experience economic difficulties after a flood, due to the dependence on the local resources. Hence, occupation in the primary economic sector is added in the vulnerability index, denoting an increase of social vulnerability.

Moreover, family structure seems to affect vulnerability distribution too. Single parent families often have lower income compared to families with two parents (Cutter et al., 2003), and have increased responsibilities regarding the assistance of other family members in case of emergency (Fitton et al., 2018). It is possible that single parent households already experience financial difficulties, which can be exacerbated during or after a hazard. Besides the struggles of evacuation, recovery process poses new difficulties since these decisions should take into consideration the wellbeing of the vulnerable family members (Fitton et al., 2018). More precisely, in a case of coastal flooding the economic cost for insurances or damage repairs might create an additional economic burden to the family. Therefore single parent families are selected for the index creation, indicating a positive relationship with vulnerability.

A final addition to the factors that affects vulnerability is social dependency. This indicator was identified by Cutter (2003) and used in the SoVI metric, but is omitted from SoVI Lite metric. However, social dependence is a crucial factor especially in the case of coastal floods. Social dependent people rely heavily on social services and governmental allowances (Cutter et al., 2003). This category includes unemployed, recipients of disability pensions and people on maternity or sickness related leaves. Therefore, people receiving public benefits might not only experience economic difficulties and struggles in the post-disaster period (Cutter et al., 2003), but they also require support during the hazard and the evacuation process. Consequently, indicators of social dependency are associated with increase of social vulnerability.

Currently, the only vulnerability assessment in regards to coastal flooding that has been undertaken in Denmark was part of the EU Flood Directive (2007/60/EC). The indicators that have been selected include population density, sights of cultural heritage, different types of infrastructure and land use (Kystdirektoratet, 2018b). These indicators are similar with the indicators of Mclaughlin and Cooper (2010). However, these variables are insufficient for measuring the social side of vulnerability, which is the aim of this research. Consequently, in order to produce a social vulnerability index for coastal floods socio-economic data for the Danish population will be used as indicators.

#### 4.2 Social Vulnerability Data

For the purposes of the analysis and the vulnerability assessment data were obtained from Statistics Denmark (2019), expect the data for Greenlandic population in Denmark that were obtained from Statistics Greenland (2018). It is common for vulnerability assessment studies to utilize census data, e.g. Cutter et al. (2003), Myers et al. (2008), Armas et al. (2016), or geodemographics, which are demographic data for commercial purposes, e.g. Willis et al. (2010) and Fitton et al. (2018). However, both census data and geodemographic were rejected for the analysis. The available census data have as basis of analysis the regions in Denmark; therefore, due to the

larger geographic area they cover, they are rejected. On the other hand, the available geodemographic databases did not correspond to the range of factors that influence vulnerability. Another unit of available data is parish level. Despite the fact that Statistics Denmark provides some data in parish level, the data are unsatisfactory for the construction of the vulnerability index. Both geodemographies and parish data only cover the population, age and gender spectrum. Therefore, for the purposes of this vulnerability assessment the spatial scale is the municipality level, as the smallest available spatial unit that covers all the factors that affect vulnerability. Currently, Denmark is divided in 98 municipalities. However, some of the collected data include Christiansø island as a municipality, while it is usually considered to belong to Bornholm municipality. Hence, Christiansø is omitted from the analysis due to data inconsistences. The majority of the obtained data are for the year 2018, with some exceptions, which can be seen in Table 4.2.

Moreover, Statistics Denmark does not provide occupation dependency data or proportion of employees in specific industries in a smaller scale than regional. Therefore, due to data unavailability, instead of the type of occupation, educational attainment of adult people was considered for the SoVI Denmark. In general, low education level is associated with higher unemployment, unstable occupations, lower income and struggles during the recovery period (Cutter et al., 2003). More precisely, people with less educational and professional qualifications are more likely to lose their job after a hazard. (Mileti, 1999; Morrow, 1999). Due to data unavailability, the indicator of Faroese population in Denmark is not taken into consideration.

Following the SoVI Lite methodology (Cutter et al., 2011; Evans et al., 2014) fourteen indicators were selected to represent the social factors that influence vulnerability (Table 4.2). Eleven indicators are considered to have a positive directional influence with vulnerability, while for three indicators (urban residents, renters and car ownership) the influence that is observed is negative (Table 4.2). A positive relationship between the indicator and the vulnerability is assigned to the indicators that increase vulnerability to coastal floods, while the negative relationship indicates a reverse effect on vulnerability. The indicators' influence is based on the literature that has been presented in the previous section (Table 4.1).

Table 4.2: The indicators that relevant to the Danish context and are utilised in the creation of the vulnerability index for Denmark. In the field of the data source the code of each dataset is presented, along with the year of the data publication. All the data are provided by Denmark's Statistics (2019), with only exception the numbers of Greenlandic people, which were obtained from Greenland's Statistics (2018).

Indicator (vulnerability relationship)	Data Source
<i>Poverty</i> : Population earning less than 60% of median income (positive) Population earning less than 50 % of median income (positive)	IFOR12A, 2017
Elderly: Population over 65 (positive)	BY2, 2019
Urban: Population of urban residents (negative)	BY3, 2019
Renters: Number of households occupied by tenant (negative)	BOLIG101, 2018
Gender: Population of females (positive)	BYA, 2029
<i>Race/ethnicity:</i> Population of Non-Western Immigrants (positive) Population of people born in Greenland and residents of Denmark (positive)	FOLK1E, 2019Q1 GDPUH, 2018

<i>Nursing population:</i> Population of people in nursing homes and home nursing (positive) Recipients of disability pension (positive)	RESIO1, 2018 HJEMSYG, 2017
Car ownership: Proportion of households with at least one vehicle (negative)	BIL600, 2017
<i>Occupation/ education:</i> Proportion of adult population who attended only basic education (positive)	HFUDD10, 2018
Social dependency: Proportion of population receiving public benefits (positive)	AUKO1, 2018Q4
Family structure: Proportion of households that consist of single parents (positive)	FAM44N, 2019

However, as Fitton et al. (2018) remark when adjusting vulnerability indicators to a specific hazard they can contribute differently than usually perceived. Often renters are perceived to have higher vulnerability, due to lower economic power. However, in the case of coastal hazards, renters might not be affected in the same way with house owners. Renters do not have to pay to rebuild or reconstruct a dwelling, neither to pay for other economic damages, since they can relocate easily. Hence, dwellings occupied by renters indicate a vulnerability decrease, or a negative directional influence, in the case of coast floods. Likewise, urban residents are more prone to a hazard, due to potential complication in the evacuation process (Cutter et al., 2000; Cutter et al., 2003; Fitton et al., 2018). Fitton et al. (2018) highlight that coastal protection measures are concentrated around areas with high population and important infrastructure. Hence, increased urban population indicates a negative influence to vulnerability for this study.

#### 4.3 Building the social vulnerability index

Once the data representing the social aspects of vulnerability are obtained, the data have to be standardized, with the use of an excel spreadsheet. Evans et al. (2014) explains that the population data should be converted into percentage values for implementing the SoVI Lite for Denmark and to allow further statistical analysis. This step is crucial because the data, shown in in Table 4.2, are in incompatible units.

The next step includes the conduction of correlation analysis. A prerequisite for the vulnerability assessment is the minimal statistical correlation among the selected vulnerability indicators (Table 4.2) (Fitton et al., 2018; Holand et al., 2011). By conducting the statistical correlation test it is confirmed that all aspects of social vulnerability are measured by a unique indicator (Fitton et al., 2018). The statistical correlation is conducted with the use of SPSS program. Based on the correlation value tier of  $\pm 0.85$ , according to Willis et al. (2010) and Fitton et al. (2018), values with higher correlation were excluded from the analysis. Based on the generated correlation matrix, two indicator were highly correlated with r=0.970. The two indicators, proportion of people with income less than 60%, and 50%, of median income, were the highly intercorrelated. Hence, the first indicator was omitted, because the two indicators explain at some extent the same information about the population.

After converting all indicators values to percentages for each municipality, it is possible to normalize the selected indicators with z-score creation. According to Evans et al. (2014) z-scores constitute a statistical normalizing method which transforms variables to a similar numeric scale. The equation for calculating z-scores is defined by Evans et al. (2014) as:

 $Z - score q = (q - \mu) / \sigma$ 

Where q is the indicator value,  $\mu$  is the mean percentage of the q values; and  $\sigma$  is the standard deviation of the q values. Therefore, mean and standard deviation of the standardized values have to be calculated for the generation of the z-scores. The calculations are conducted in an excel spreadsheet.

Once the SoVI scores for each municipality are generated, the scores are imported in ArcGIS 10.6. With the use of choropleth mapping of the standard deviations of the scores, social vulnerability to coastal floods is illustrated spatially. Emrich and Cutter (2011) classified vulnerability scores as limited, moderate and elevated, while the same classification method was used by Evans et al. (2014). Borden et al. (2008) utilize a classification tier of 0.5 standard deviation as the breaking point among the classes mentioned above. In this research, vulnerability scores are classified following the tier of 0.5 in order to depict the vulnerability scores.

SoVI Lite method has been the basis of the vulnerability assessment. As it has been described before, this method does not impose a PCA for the indicators selection and the index composition. Hence, for the purposes of this research PCA is constructed after the index composition in order to identify the main drivers of social vulnerability in Denmark.
# **5** Results

One the social vulnerability index for coastal flooding in Denmark is constructed, it is possible to assess the vulnerability of each municipality. Therefore, this chapter presents the results of the conducted research and elaborates on the main factors of social vulnerability across the country.

### 5.1 Where do vulnerable people live?

It might not be surprising that the vast majority of the Danish municipalities expose moderate levels of vulnerability. The SoVI scores for the 98 Danish municipalities fluctuate from +3.33 (the highest value) to -1.43 (the lowest value). Overall, thirteen municipalities scored as very highly vulnerable with standard deviation of the total SoVI scores higher than +1.0, and seven municipalities as highly vulnerable. Forty one municipalities, representing the majority, have moderate vulnerability levels. Twenty seven Danish municipalities have low vulnerability, while nine of the municipalities scored very low on social vulnerability, with scores less than -1.0. The key findings can be seen in Table 5.1.

Table 5.1: The municipalities with the highest and the lowest vulnerability scores. The municipalities are classified from the largest score to the smallest in the first column, while the second column uses the reverse classification.

Most Vulnerable Municipalities	Least Vulnerable Municipalities
Ærø	Ballerup
Aarhus	Brønderslev
Odense	Herlev
Læsø	Varde
Copenhagen	Glostrup
Aalborg	Helsingør
Assens	Faxe
Samsø	Faaborg-Midtfyn
Fanø	Ringkøbing-Skjern
Lolland	Vesthimmerlands

One cluster of the most vulnerable people can be found in the urban areas of the country. Some of the highest vulnerable scores are attributed to Copenhagen, Aarhus, Odense and Aalborg, while similar vulnerability levels are noticed in Copenhagen's neighboring municipalities. The four biggest cities in Denmark are classified among the top ten most vulnerable municipalities across the country (Table 5.1). These urbanized areas share similar demographic traits, such as the highest population concentration in the country and the greater proportion of Non-Western citizens. Moreover, they constitute of highly social dependent and low income population, and the majority of the citizens are tenants. Thus, the most vulnerable population is not equally distributed among the urban centers of Denmark (Figure 5.1). On the contrary, among the most vulnerable municipalities are inhabited islands, like Æro, Samsø and Læsø (Figure 5.1), which are placed among the top five most vulnerable municipalities. It is observed from the data, that vulnerability in this case is shaped by the high proportion of elderly and their associated dependence due to health issues. The islands of Denmark have increased population in nursing facilities or home nursing and high proportion of recipients of disability pension. Moreover the islands have scored high in the primary education attainment, while the mean income of the residents is 60% lower than the average individual income. Moreover, these areas scored high on the concentration of Greenlandic

population, in a similar way with Aalborg and Odense. This is not surprising when considering the social challenges that rural Denmark is facing (Andersson & Laursen, 2011; Laursen, 2009). The rural parts of Denmark suffer from out-migration of young and skilled citizens, slow economic growth and high dependency on local resources and the primary sector (Andersson & Laursen, 2011).



Figure 5.1: Social vulnerability of Danish municipalities. The municipalities around the metropolitan area of Copenhagen are enlarged to provide a detailed view due to the concentration of small municipalities. The municipalities are ranked based on the standard deviation of the derived vulnerability scores, and are classified according to the legend of the map.

Municipalities that have scored less than -1.0 are labeled as very low vulnerability, and they are distributed unevenly across the country (Figure 5.1). Faaborg-Midfyn, Faxe, Varde and Brønderslev municipalities are among the areas where the least vulnerable live (Table 5.1). The majority of the low vulnerable municipalities include a mix of low urbanized municipalities, with low proportion of economic disadvantaged and social dependent citizens. The least social vulnerable people live in Ballerup municipality, while there is at least one municipality labeled as very low vulnerable in each region. In general, low vulnerability is the result of low social challenges. Low vulnerable municipalities are spread across the country, with a big cluster in the middle of Jutland (Figure 5.1). With similar population composition as the very low vulnerable municipalities, the majority of the population is mainly Danish, educated and owners of at least one vehicle.

The remaining municipalities are labeled as moderate vulnerable (Figure 5.1). This category includes forty one Danish municipalities. These areas present a relatively similarity in the social and economic situation of the population. They are rural and sub-urban areas, with limited ethnic background of the citizens, who are qualified for the labor market and with income higher than the average. Overall, the moderate scores of vulnerability are generated through the interaction of factors that increase and decrease the total vulnerability. Some notable clusters of moderate vulnerable municipalities can be observed in close proximity to the bigger urban centers of the country, e.g Aarhus, Odense and Copenhagen, and in the southern part of Jutland region.

#### 5.2 Which are the main factors of social vulnerability?

For a holistic interpretation of the SoVI scores, PCA is conducted in order to highlight the most influential factors of the social vulnerability to coastal floods in Demark. While PCA is usually applied in order to identify the factors influencing vulnerability among a large set of variables and facilitate the index construction, for the purposes of this research PCA is conducted as a validation measure. The overall aim with the use of this method is to reassure that the selected indicators that are utilised to create the social vulnerability index for coastal floods in Denmark can be identified with the more traditional use of the PCA.

Once the PCA was conducted, three components are extracted with eigenvalues more than + 1.0 (Figure 5.2). The cumulative variance of these three components is around 83%, a percentage in accordance to the results of Cutter et al. (2011; 2003). In other words, these three cumulative components contain almost 83% of the information contained in the original set of variables. Therefore, it can be argued that the selected socio-economic data that have been used for the Danish vulnerability index are appropriate for conceptualizing the vulnerabilities of the population.



Figure 5.2: Scree plot diagram of the derived principal components. The three components have eigenvalues greater than +1.0 and are the components that contains the greatest amount of information of the original indicator values. The three components are named *Age, gender and socioeconomic status, Immigrants, Urbanism and special needs, and Accessibility and ethnicity.* The individual indicators of each component are explained further within the chapter.

The first component, named *Age, gender and socioeconomic status* (Figure 5.3) explains 62 % of the variance. This component distinguishes the municipalities based on the proportion of elderly, tenants, people with low income and dependent on social funding, female population, education level and family structure. It can be seen that the most vulnerable municipalities have scored high vulnerability as individual indicators of the first component (Figure 5.3). More precisely, the vulnerability of the urban municipalities seems to be related with high scores in tenants, income and the people receiving public benefit. On the contrary, the rural municipalities that have labeled as high and very high vulnerable can be attributed to increased scores in primary education, age, gender and single parents' families.

The second component explains 11 % of the total variance comprising of indicators about the proportion of nonwestern immigrants and their descendants, recipients of disability pension and urban population (Figure 5.4). According to Figure 5.4, the proportion of people receiving disability pension is moderate across the country, with a notable exception in Funen Island. Moreover, the immigration scores correspond to the total vulnerability patterns, besides the Western part of Zealand.

The final component accounts for 10 % of the total variance. This component comprises of the individual scores of car ownership across the county, as well as the proportion of Greenlandic population. Figure 5.5 shows that high concentration of Greenlandic population does not imply high vulnerability, while cart ownership scores seem to correspond with the total vulnerability scores, especially in the moderate vulnerable municipalities.



Figure 5.3: Vulnerability indicators that comprise component one. The individual indicators are presented based on their loadings classified from the largest to the smallest. The loadings of the indicators can be seen in Appendix



Figure 5.4: Component number two. The individual indicators are presented based on their loadings from the largest to the smallest. The loadings of the indicators can be seen in Appendix



Figure 5.5: Component number three. The individual indicators are presented based on their loadings from the largest to the smallest. The loadings of the indicators can be seen in Appendix

## 6 Discussion

In section 1.7 the main objectives of this research were declared. These objectives are the following:

- Where do vulnerable people live in Denmark?
- What are the spatial patterns between and among the Danish municipalities?

This chapter aims at interpreting and reflecting upon the results reported in Chapter 5 and to discuss the implications of the study for coastal flood risk management in Denmark. Additionally, the limitations of the study are discussed, followed by suggestions for future research.

## 6.1 Key findings

The first aim is locating the most vulnerable population in Denmark. This has been achieved by creating the Social Vulnerability Index to coastal flooding for 97 Danish Municipalities, as it can be seen in chapter 5. According to the findings of this research, the most vulnerable people in Denmark are located in the four biggest cities and in the islands across the inner coast of the country (Table 5.1). The second aim of the research is captured by the second research question. With the use of the social vulnerability map for Denmark (Figure 6.1), it is possible to visualise the vulnerability scores of each municipality, allowing detecting patterns between the municipalities. Moreover, in section 5.2, the individual drivers of vulnerability are mapped in order to provide holistic view of the factors of that affect vulnerability in each municipality. Two cluster of very high social vulnerability are observed in Denmark. The first cluster comprises of the highly urbanized municipalities, which include large Non-Western immigrants' populations and single-parent families, low income populations and large numbers of people dependent on public benefits. The second cluster of highly vulnerability contains the islands of the inner coast of the county, comprising by elderly population, low skilled labor force and socially dependent population (public benefits, disability pension, single parents families, nursing). Another vulnerability pattern that is observed refers to the least vulnerable people, who reside in the suburbs of Copenhagen (Figure 6.1). A pattern of low vulnerability can be seen in Figure 6.1, where some of least vulnerable municipalities cluster. The municipalities include educated and wealthy populations, as well as families, factors that decrease social vulnerability.

The patterns of vulnerability suggest that social vulnerability of the Danish population, especially in the case of the high vulnerable islands, could be explained by the spatial polarization in the country. Andersson and Laursen (2011) explain that in the last decades the rural parts of the country struggle with out-migration of the young and skilled people and experience economic decline, due to centralization of economic opportunities in bigger cities. While, Hansen and Smidt-Jensen (2004) suggest that the lack of infrastructure and of skilled population can explain the decline in the rural parts of Denmark. The analysis of the individual vulnerability components propose that the high vulnerability of islands is influenced by the age and the educational level of the population. On the contrary, the highly urbanized municipalities of Denmark are experiencing economic and population growth (Andersson & Laursen, 2011), which explains the high concentration of skilled people. However, due to large population concentration, it is not surprising that the big cities include high numbers of marginalized populations (e.g. Non-Western immigrants, recipients of public benefits), which are stipulated to increase social vulnerability. Therefore, the results of the study propose that the spatial variance of social vulnerability to coastal floods in influenced at some extend by the urban development tendencies across Denmark.

Due to the different scale of analysis, the results of the generated vulnerability assessment to coastal floods cannot be compared with the vulnerability assessment of the DCA (Figure 6.1). However, it can be observed that both maps highlight the vulnerability of areas with increased population, such as Copenhagen, Aarhus etc. Nevertheless, the vulnerability assessment of the DCA differs with the generated assessment in the vulnerability of the islands along the inner coast of the country. As it is argued in Chapter 1, the existing vulnerability assessment is not an assessment of the vulnerabilities of the Danish population to coastal floods, since it includes indicators such as critical infrastructure, local economic activity and population density (Table 1.2). Therefore a reason behind the difference of the observed vulnerability patterns between the two maps (Figure 6.1) could be that the previous vulnerability assessment of the coastal floods underrepresents the vulnerability of rural areas because of the absence of critical infrastructure and the lower population.



Figure 6.1: Map of social vulnerability to coastal floods generated for the purposes of this reasearch, and vulnerability map that was created by the DCA (2018a).

#### 6.2 Implications for flood risk management in Denmark

It should be clear by now that flood risk assessment requires knowledge of the local conditions and the vulnerabilities of local population (Jebens & Sorensen, 2017). Sorensen and Jebens (2015) explain that the vulnerability assessment that was implemented by the DCA aimed at identifying social vulnerabilities. Thus, as it is discussed in section 6.1, this goal has not been achieved. Despite the identification of prone areas to flooding across the country, without the assessment of the social aspects of vulnerability, the results of the process can be misleading. The methodology that has been followed could either be incorporated into the existing flood risk assessment as a tool for identifying the social vulnerabilities of the society, which is missing. Alternatively, it may serve as the theoretical basis of a new way of defining hazard, risk and vulnerability that could lead in a holistic flood risk assessments and the establishment of nation flood programme.

The interview with Vejle's Resilience officer highlights the confusion around implementing social-oriented vulnerability or adaptation plans, as well as quantifying social vulnerability. She further discusses that:

"For instance, let's say that the entire city of Vejle gets totally flooded and people lose their homes. Are all of them insured? Will the insurance companies help them all? We don't have any knowledge about something like that, because in Denmark insurance is a private matter not a public one. Therefore, we have no idea how many people are insured. For deeper understanding of we have to take into consideration the whole community and most of all the vulnerable groups."

This quote supports the argument that awareness regarding the threat of coastal floods and their socio-economic impacts is limited in Denmark (Sorensen & Jebens, 2015). The flood management implementation is a municipal responsibility, while according to the Danish Coastal Act individuals are responsible for protecting themselves and their properties against flooding (Jebens & Sorensen, 2017). Thus, Danish society tends to undermine their vulnerability to coastal floods and this perception contributes to a superficial understanding of the threat posed by coastal floods. This study has generated maps of the individual drivers of vulnerability that can be used in the municipal adaption plans and the local flood risk management. Therefore, the results of this research can assist municipalities in the identification of their vulnerabilities based on their population composition and their needs. Consequently, local planning can re oriented towards the needs of the most vulnerable populations. At the same time, these results can increase the awareness of citizens, relevant stakeholders and emergency managements. Additionally, the greater awareness of the social vulnerabilities can promote engagement of the vulnerable groups in the flood risk management and the local planning process, which is an aim of many municipalities.

Jebens and Sorensen (2017) further discuss that the coping with floods would pose a challenge for Denmark due to problems of communication and coordination among stakeholders. Hence, this research can act as a barrier around the confusion of social vulnerabilities understanding among different stakeholders. Overall, this thesis contributes to the body of knowledge in the coastal flood risk management and vulnerability field. The findings of the research emphasise the applicability and transferability of the SoVI Lite methodology to different geographic contexts, while it comprises a step towards the construction of an index tailored to the Danish social context. This study exposes the social vulnerabilities of the Danish population, which is an important step for a mentality change towards vulnerability and more effective coastal flood risk management.

## 6.3 Limitations of the study

Social vulnerability assessments do not provide "end-point" results (Solangaarachchi, et al., 2012). By reflecting on the way this this research defines social vulnerability (Chapter 3), readers should keep in mind that social vulnerability is a component of the place vulnerability of each municipality. The hazard of place model dictates that social and biophysical vulnerability interact in order to create the place vulnerability. It is acknowledged by the authors that a complete vulnerability assessment is a long term project. Hence, this research contributes to the assessment only of the social aspects of vulnerability to coastal flooding.

Another challenge during this research was the potential bias during the indicators selection process. In order to address this limitation, extensive literature search was conducted and the selected indicators correspond to the general factors that are broadly identified as important in vulnerability assessments. Thus, challenges related to available data interfere with the vulnerability assessment. Moreover, the social vulnerability is conducted with

municipalities as the unit of the analysis, because they constitute the smallest administrative unit of analysis with plethora of available data. It is acknowledged that data at municipal level do not provide the greatest spatial resolution for the purposes of this research. Ideally, parish, postal code or address level data would offer a better resolution and present in further detail the spatial distribution of social vulnerability. However, data unavailability influences not only the indicators, but also the unit of the analysis. As a consequence, municipalities were selected since they comprise an acceptable unit of analysis for achieving the aims of this research.

A generalized limitation of the vulnerability indices is the conceptualization the dynamic processes that shape vulnerability (Schumann & Moura, 2015). Fekete (2012) elaborates on this challenge and explains that indicators can be used partially for interpreting a phenomenon. Rygel et al. (2006) argues that interpretation of experiences, ideas and cultural values cannot easily be represented by quantitative data. Therefore a small set of indicators has to be collected, comprising of the indicators that conceptualize the geographic context in the best way possible, in order to quantify vulnerability (Fekete, 2012). Adger et al. (2004) agree that developing of a vulnerability index is important for allowing comparison among places. Nevertheless, vulnerability indices are associated sometimes with over- or under-presentation of the individual factors, since the combined vulnerability score has the potential to favor or not individual factors, and hence lead to misinterpretation of the findings (Rygel et al., 2006). Since SoVI provides relative vulnerability scores (Cutter et al., 2003; Rygel et al., 2006), visualisation of the individual indicators is selected in order to provide a different perspective of the spatial distribution of vulnerability drivers. In this way, it is possible to draw the attention into the specific challenges that each area faces, instead of focusing solely on the generated vulnerability score. Another way to address this limitation is the validation of the results based on secondary disaster data. However, validation is omitted due data unavailability for coastal floods in Denmark, strengthening the argument that validation is a difficult research obstacle (Adger et al., 2004; Fekete, 2009; Fekete, 2012).

### 6.4 Future research recommendations

Within this research knowledge gaps are identified, which require to be addressed by future research in order to improve coastal flood management in Denmark. The suggestions include:

- Use of the original SoVI methodology, instead of SoVI Lite as the analytical framework of the social vulnerability assessment. By including more data, different patterns of vulnerability can be observed and the drivers of vulnerability can be explored in-depth;
- Use of data with greater spatial resolution. The analysis should be local, since the flood management plans are implemented by municipalities, but socio-economic data at parish, postal code or even address level will contribute to a better understanding of the spatial distribution of vulnerabilities;
- Investigation of past hazard events, both with quantitative and qualitative methods, which can create data set for validating the results of the social vulnerability index for Denmark. In this way, the complex social interactions, the historical and cultural drivers of vulnerability would be taken into consideration. The validation of the model's results can enhance the possibilities that the method poses for stakeholders across the country, while it can be applied to other types of hazards;
- Assessment of the future social vulnerabilities with projections of data and scenario building as a tool.

## 7 Conclusion

The conducted research focuses on the spatial distribution of the social vulnerability to coastal flooding among the Danish municipalities. More precisely the research aims at quantifying the social vulnerabilities in order to identify the most vulnerable populations and understand their social characteristics, by stating the two following research questions:

#### • Where do vulnerable people live in Denmark?

#### • What are the spatial patterns between and among the Danish municipalities?

By implementing the SoVI Lite methodology, which has been developed by Cutter et al. (2011), and the use of socio-economic data at municipal level, a social vulnerability index for coastal flooding is constructed for Denmark. The index comprises of 12 indicators and it has been used for assessing the vulnerability to coastal floods in of 97 Danish municipalities. Each municipality is assigned a vulnerability score as a whole unit. Therefore, the first aim of this research has been fully achieved with the creation of the social vulnerability index and the quantification of the vulnerability of each municipalities, almost 42%, demonstrate moderate levels of social vulnerability. A considerable lower number of municipalities, 37%, are labeled as low vulnerable. Overall, 20 municipalities (21%) have been assigned high vulnerability to coastal flooding.

Most Vulnerable Municipalities	Least Vulnerable Municipalities
Ærø	Ballerup
Aarhus	Brønderslev
Odense	Herlev
Læsø	Varde
Copenhagen	Glostrup
Aalborg	Helsingør
Assens	Faxe
Samsø	Faaborg-Midtfyn
Fanø	Ringkøbing-Skjern
Lolland	Vesthimmerlands

Table 7.1: Key findings

Among the ten most vulnerable municipalities across the country, the four biggest urban centers of Denmark can be found. The top ten ranking also includes Læsø, Samsø, Fanø, Æro and Lolland, some of the big inhabited islands of the country (Figure 6.1). Once the vulnerability was quantified, it was possible to investigate the drivers of vulnerability in each municipality. The results of this study propose the existence of two contradictory spatial patterns among the vulnerable municipalities of Denmark. One notable cluster of high vulnerability includes the urban municipalities, while the second cluster of high vulnerability comprises of the rural islands of the inner coast. These two clusters can be attributed to contradictory factors, since the population composition is different among the areas.

In order to accomplish the second aim of this research PCA is conducted to highlight the most influential factor of the social vulnerability in Demark. The results of the PCA point out three components as the major drivers of social vulnerability in terms of coastal floods. The components are named *Age, gender and socioeconomic status, Immigrants, disability and urbanism, and Ethnicity and accessibility* respectively, and explain 83% of the data variance. The results indicate that the high vulnerability of the rural municipalities is connected with the demographic challenges that these parts of the county face. These municipalities are comprised mainly by elderly, low income or social dependent population, factors that affect positively the vulnerability and can explain the high levels of it. On the contrary, the most urbanized areas of the country, such as Copenhagen and Aarhus, are characterised by large numbers of immigrants and people dependent on financial support by the Danish state, which increase the vulnerability of urban municipalities. Therefore, the findings of the research suggest that social vulnerability to coastal floods vary among the Danish municipalities and these patterns follow at some extent the development of the country.

This resrearch contributes to the understanding of the social factors that can interfere with the ability of the Danish population to cope with hazard when encountered a coastal flooding. The findings of the research could either be incorporated into the existing flood risk assessment, serving as a tool for identifying the social vulnerabilities of the society, which is missing. Moreover, the identification of the most vulnerable municipalities can facilitate the planning process and the implementation of flood risk plans, considering the responsibility of municipalities to implement these plans. Municipalities can use the results of the study to understand their main social challenges and raise awareness among the civil society. As coastal floods are expected to become a great threat for coastal Denmark, social vulnerability assessment should be part of the coastal flood management process.

Further research is required for overcoming the challenges of creating an index tailored to the Danish context, while incorporating both qualitative and quantitative data. In this way, the complexities of the Danish population, the historical and cultural drivers of vulnerability will be taken into consideration. While, further research can propose ways to validate the results of the social vulnerability index for the coastal floods in Denamrk, in order to can enhance the possibilities that the index poses.

## 8 Bibliography

- Adger, W. N., Brooks, N., Bentham, G., & Agnew, M. (2004). New indicators of vulnerability and adaptive capacity. *Tyndall Centre for Climate Change Research*, 5(January), 128. https://doi.org/10.1016/j.cej.2004.12.010
- Adger, W. N. (2006a). Vulnerability. *Global Environmental Change*, 16(3), 268–281. https://doi.org/10.1016/j.gloenvcha.2006.02.006
- Adger, W. N. (2006b). Vulnerability. *Global Environmental Change*, 16(3), 268–281. https://doi.org/10.1016/j.gloenvcha.2006.02.006
- Agence Européenne de l'Environnement, E. (2013). Flood risk in Europe: the long-term outlook. Retrieved from http://www.eea.europa.eu/highlights/flood-risk-in-europe-2013
- Alexander, D. (2000). Confronting Catastrophe. New Perspectives on Natural Disasters.
- Andersson, H. R., & Laursen, L. L. (2011). Differentiated decline in Danish outskirt areas. *Landinspektoeren*, 46(1), 96–113.
- Armaş, I., & Gavriş, A. (2016). Census-based Social Vulnerability Assessment for Bucharest. Procedia Environmental Sciences, 32, 138–146. https://doi.org/10.1016/j.proenv.2016.03.018
- Bernard, H. R. (2006). Interviewing: Unstructured and Semi-structured. *Research Methods in Anthropology: Qualitative and Quantitative Approaches, 4th, 201–250.* Retrieved from http://www.loc.gov/catdir/toc/ecip0515/2005018836.html
- Bianchi, S. M., & Spain, D. (1996). Women, work, and family in America. *Population Bulletin, Population Reference Bureau*, 51(3).
- Birkmann, J., Cardona, O. D., Carreño, M. L., Barbat, A. H., Pelling, M., Schneiderbauer, S., ... Welle, T. (2013). Framing vulnerability, risk and societal responses: The MOVE framework. *Natural Hazards*, 67(2), 193–211. https://doi.org/10.1007/s11069-013-0558-5
- Birkmann, J. (2005). Danger Need Not Spell Disaster But How Vulnerable Are We? Research Brief. United Nations University, Tokyo., (1), 1–8.
- Birkmann, J. (2006a). Basic principles and theoretical basis (Vol. 01, pp. 7–54). https://doi.org/10.1016/j.compedu.2010.02.018
- Birkmann, J. (2006b). Measuring vulnerability to promote disaster-resilient societies : Conceptual frameworks and definitions. *Measuring Vulnerability to Natural Hazards*, 01, 9–54. https://doi.org/10.1111/j.1539-6975.2010.01389.x
- Blaikie, P., Cannon, T., Davis, I., & Wisner, B. (1994). At Risk: Natural Hazards, People's Vulnerability and Disasters. London: Routledge.
- Bohle, H.-G. (2001). Vulnerability and Criticality: Perspectives from Social Geography. Update IHDP 2/01. Newsletter of the International Human Dimensions Programme on Global Environmental Change, (2), 1– 7. Retrieved from http://scholar.google.com/scholar?hl=en&btnG=Search&q=intitle:Vulnerability+and+Criticality:+Perspecti

ves+from+Social+Geography#0

Boligministeriet, T.-B. (2017). Ghettoområder.

- Bollin, C., Cárdenas, C., Hahn, H., & Vatsa, K. S. (2003). Natural Disasters Network Disaster Risk Management by Communities and local Governments.
- Borden, K. A., & Cutter, S. L. (2008). Spatial patterns of natural hazards mortality in the United States. *International Journal of Health Geographics*, 7, 1–13. https://doi.org/10.1186/1476-072X-7-64
- Bowen, G. A. (2009). Document Analysis as a Qualitative Research Method. *Qualitative Research Journal*. https://doi.org/10.3316/QRJ0902027
- Breil, M., Cmcc, F., Ukcip, C. D., Kazmierczak, A., Syke, K. M., Ft, L. R., ... Wer, R. S. (2018). Social vulnerability to climate change in European cities state of play in policy and practice Contributors : *European Topic Centre on Climate Change Impacts, Vulnerability and Adaptation (ETC/CCA) Technical Paper* 2018/1, (Fondazione CMCC), 1–86. https://doi.org/10.25424/CMCC/SOCVUL EUROPCITIES
- Brooks, N. (2003). Vulnerability, risk and adaptation: A conceptual framework (No. Working paper 38,20).
- Bruun, H. H., Ejrnes, R., Buttenschøn, R. M., Camille, S., Damgaard, C., Ejrnæs, R., ... Nordjylland, R. (2012). *Arealanvendelse i Danmark siden slutningen af 1800-tallet. Faglig rapport fra DMU* (Vol. 16). https://doi.org/10.1017/CBO9781107415324.004
- Burton, C., Rufat, S., & Tate, E. (2018). Social Vulnerability: Conceptual Foundations and Geospatial Modeling, (53–81). Retrieved from https://hal.archives-ouvertes.fr/hal-02001407/
- Burton, I., Kates, R., & White, G. (1978). The Environment as Hazard. (T. G. Press, Ed.). New York, NY.
- Cardona, Omar D. (2003). The Need for Rethinking the Concepts of Vulnerability and Risk from a Holistic Perspective : A Necessary Review and Criticism for Effective. In *Development*.
- Cardona, Omar Dario, Van Aalst, M. K., Birkmann, J., Fordham, M., Mc Gregor, G., Rosa, P., ... Thomalla, F. (2012). Determinants of risk: Exposure and vulnerability. In *Managing the Risks of Extreme Events and Disasters to Advance Climate Change Adaptation: Special Report of the Intergovernmental Panel on Climate Change*. https://doi.org/10.1017/CBO9781139177245.005
- Chambers, R. (1989). Vulnerability, coping and policy. *IDS Bulletin. Institute of Development Studies*, *37*(2), 33. https://doi.org/10.1111/j.1759-5436.2006.tb00284.x
- Chen, W., Cutter, S. L., Emrich, C. T., & Shi, P. (2013). Measuring social vulnerability to natural hazards in the Yangtze River Delta region, China. *International Journal of Disaster Risk Science*, 4(4), 169–181. https://doi.org/10.1007/s13753-013-0018-6
- Chipangura, P., Dewald Van Niekerk, & Waldt, G. Van Der. (2016). An exploration of objectivism and social constructivism within the context of disaster risk. *Disaster Prevention and Management*, 25(2), 261–274. https://doi.org/http://dx.doi.org/10.1108/MRR-09-2015-0216
- Chorynski, A., Ponskwar, I., Kron, W., Brakenridge, G., & Kundzewicz, Z. W. (2012). Catalogue of Large Floods in Europe in the 20th Century. In *Changes in Flood Risk in Europe*. https://doi.org/10.1201/b12348-5

- Clark, G. E., Moser, S. C., Ratick, S. J., Dow, K., Meyer, W. B., Emani, S., ... Schwarz, H. E. (1998). Assessing the vulnerability of coastal communities to extreme storms: The case of reverse, MA., USA. *Mitigation and Adaptation Strategies for Global Change*. https://doi.org/10.1023/A:1009609710795
- Clifford, N., French, S., & Valentine, G. (2010). Key Methods in Geography. SAGE Publications. https://doi.org/10.1017/CBO9781107415324.004
- Cook, J., Oreskes, N., Doran, P. T., Anderegg, W. R. L., Verheggen, B., Maibach, E. W., ... Rice, K. (2016). Consensus on consensus: a synthesis of consensus estimates on human-caused global warming. *Environmental Research Letters*, 11(4), 048002. https://doi.org/10.1088/1748-9326/11/4/048002
- Costa, L., & Kropp, J. P. (2013). Linking components of vulnerability in theoretic frameworks and case studies. Sustainability Science, 8(1), 1–9. https://doi.org/10.1007/s11625-012-0158-4
- Cova, T. J., & Church, R. L. (1997). Modelling community evacuation vulnerability using GIS. *International Journal of Geographical Information Science*. https://doi.org/10.1080/136588197242077
- Creswell, J. W. (2009). Research design: qualitative, quantitative and mixed approaches. Research Design. https://doi.org/10.2307/1523157
- Creswell, J. W. (2013). Qualitative Inquiry & Research Design: Choosing Among Five Approaches. Sage Publications. https://doi.org/10.1111/1467-9299.00177
- Cutter, S., Emrich, C., Webb, J., & Morath, D. (2009). Social Vulnerability to Climate Variability Hazards: A Review of the Literature. *Hazards and Vulnerability Research Institute*, 1–44.
- Cutter, S. L., & Finch, C. (2008). Temporal and spatial changes in social vulnerability to natural hazards. *Proceedings of the National Academy of Sciences*, 105(7), 2301–2306. https://doi.org/10.1073/pnas.0710375105
- Cutter, S. L., Barnes, L., Berry, M., Burton, C., Evans, E., Tate, E., & Webb, J. (2008). A place-based model for understanding community resilience to natural disasters. *Global Environmental Change*, *18*(4), 598–606. https://doi.org/10.1016/j.gloenvcha.2008.07.013
- Cutter, S. L., & Emrich, C. T. (2006). Moral Hazard, Social Catastrophe: The Changing Face of Vulnerability along the Hurricane Coasts. *The ANNALS of the American Academy of Political and Social Science*, 604(1), 102–112. https://doi.org/10.1177/0002716205285515
- Cutter, S. L., Emrich, C. T., & Morath, D. (2011). Social vulnerability and place vulnerability analysis methods and application for Corps planning: Technical analyses. In M. C. Dunning & S. Durden (Eds.), *In Social Vulnerability Analysis Methods for Corps Planning* (pp. 74–88). Institute for Water Resources: U.S. Army Corps of Engineers.
- Cutter, S. L., Mitchell, J. T., & Scott, M. S. (2000). Revealing the vulnerability of people and places: A case study of georgetown county, South Carolina. *Annals of the Association of American Geographers*. https://doi.org/10.1111/0004-5608.00219
- Cutter, S. L. (1996). Vulnerability to hazards. *Progress in Human Geography*, 20(4), 529–539. https://doi.org/10.1177/030913259602000407
- Cutter, S. L, Boruff, B. J., & Shirley, W. L. (2003). Social Vulnerability to Environmental Hazards, 84(2), 242–261.

Danish Government. (2012). How to manage cloudburst and rain water, (December).

Danmarks Statistik. (n.d.). Danmarks Statistik. Retrieved May 16, 2019, from https://www.dst.dk/da

Danmarks Statistik. (2019). Danmarks Statistik. Retrieved June 4, 2019, from https://www.dst.dk/da/

- Davidson, R. A., & Shah, H. C. (1997). An Urban Earthquake Disaster Risk Index. The John A. Blume Earthquake Engineering Center, Department of Civil and Environmental Engineering, Report No. 121.
- Dilley, M., Chen, R. S., Deichmann, U., Lerner-Lam, A. L., & Arnold, M. (2005). Natural Disaster Hotspots. Natural Disaster Hotspots. https://doi.org/10.1596/0-8213-5930-4
- Doran, P. T., & Zimmerman, M. K. (2009). Examining the Scientific Consensus on Climate Change. *Eos, Transactions American Geophysical Union*, 90(3), 22. https://doi.org/10.1029/2009EO030002
- Downing, T. E., Butterfield, R., Cohen, S., Huq, S., Moss, R., Rahman, A., ... Stephen, L. (2001). Climate change vulnerability: linking impacts and adaptation.
- Dunning, C. M., & Durden, S. (2011). Social Vulnerability Analysis Methods for Corps Planning. *Institute for Water Resources Publications*, (May), 105. Retrieved from http://www.iwr.usace.army.mil/Portals/70/docs/iwrreports/2011-R-07.pdf
- Dunning, M. C., & Durden, S. (2013). Social Vulnerability Analysis: A Comparison of Tools. *IWR White Paper*, (February), 1–34. https://doi.org/10.5194/nhess-10-1697-2010
- Dunteman, G. H. (1989). Principal Components Analysis. In *Quantitative Applications in the Social Sciences*. SAGE.
- Dwyer, A., Zoppou, C., & Nielsen, O. (2004). Quantifying Social Vulnerability: A methodology for identifying those at risk to natural hazards. *Geoscience Australia*. https://doi.org/ISBN: 1 920871 09 8
- Eakin, H., & Luers, A. L. (2006). Assessing the Vulnerability of Social-Environmental Systems. *Annual Review* of Environment and Resources, 31(1), 365–394. https://doi.org/10.1146/annurev.energy.30.050504.144352
- EEA. (2017). Climate change adaptation and disaster risk reduction in Europe. Enhancing coherence of the knowledge base, policies and practices European Environment Agency. EEA Report. https://doi.org/10.1007/s13398-014-0173-7.2
- Emrich, C. T., & Cutter, S. L. (2011). Social Vulnerability to Climate-Sensitive Hazards in the Southern United States. Weather, Climate, and Society, 3(3), 193–208. https://doi.org/10.1175/2011WCAS1092.1
- Enarson, E, & Morrow, B. H. (1997). A gendered perspective: The voices of women. In W G Peacock, B. H. Morrow, & H. Gladwin (Eds.), *Hurricane Andrew: Ethnicity, Gender, and the Sociology of Disasters* (pp. 116–140.). Miami, FLorida: International Hurricane Center, Laboratory for Social and Behavioral Research.
- Enarson, Elaine, & Scanlon, J. (2002). Gender patterns in flood evacuation: A case study in Canada's Red River Valley. *Applied Behavioral Science Review*. https://doi.org/10.1016/s1068-8595(00)80013-6
- European Environment Agency. (2011). Disasters in Europe: more frequent and causing more damage. https://doi.org/10.1007/978-981-13-0974-8

- Evans, J., Hauer, M. E., Hardy, D., & Pippin, J. S. (2014). Assessing social vulnerability using SoVI-Lite: A demonstration study at Glynn County, GA, (January 2015). https://doi.org/10.13140/2.1.2329.3765
- Fekete, A. (2009). Validation of a social vulnerability index in context to river-floods in Germany. *Natural Hazards and Earth System Science*, 9(2), 393–403. https://doi.org/10.5194/nhess-9-393-2009
- Fekete, Alexander. (2012). Spatial disaster vulnerability and risk assessments: Challenges in their quality and acceptance. *Natural Hazards*, 61(3), 1161–1178. https://doi.org/10.1007/s11069-011-9973-7
- Fekete, Alexander, Damm, M., & Birkmann, J. (2010). Scales as a challenge for vulnerability assessment. *Natural Hazards*, 55(3), 729–747. https://doi.org/10.1007/s11069-009-9445-5
- Fink, A. (2005). Conducting research literature reviews : from the Internet to paper (Second edi). Thousands Oaks, CA: SAGE.
- Fitton, J. M., Hansom, J. D., & Rennie, A. F. (2018). A method for modelling coastal erosion risk: the example of Scotland. *Natural Hazards*, 91(3), 931–961. https://doi.org/10.1007/s11069-017-3164-0
- Fothergill, A. (1996). Gender, Risk, and Disaster. International Journal of Mass Emergencies and Disasters.
- Fothergill, A., & Peek, L. A. (2004). Poverty and disasters in the United States: A review of recent sociological findings. *Natural Hazards*. https://doi.org/10.1023/B:NHAZ.0000026792.76181.d9
- Füssel, H. M. (2007). Vulnerability: A generally applicable conceptual framework for climate change research. Global Environmental Change, 17(2), 155–167. https://doi.org/10.1016/j.gloenvcha.2006.05.002
- Füssel, H. M., & Klein, R. J. T. (2006). Climate change vulnerability assessments: An evolution of conceptual thinking. *Climatic Change*, 75(3), 301–329. https://doi.org/10.1007/s10584-006-0329-3
- Goodsite, M.E., M. Davis, R.J.T. Klein, B. Davídsdóttir, R. Atlason S. Juhola, M. Landauer, B.O. Linnér, T. Neset, E. Glaas, G. E. (2013). White Paper: Climate Change Adaptation in the Nordic Countries, 40. Retrieved from http://www.sei-international.org/publications?pid=2275
- Grønlands Statistik. (2018). Personer født i Grønland bosiddende i Danmark. Retrieved June 4, 2019, from http://www.stat.gl/dialog/main.asp?lang=da&version=201901&sc=GD&subthemecode=O1&colcode=O
- Grothmann, T., & Reusswig, F. (2006). People at risk of flooding: Why some residents take precautionary action while others do not. *Natural Hazards*, *38*(1–2), 101–120. https://doi.org/10.1007/s11069-005-8604-6
- Guillard-Goncąlves, C., Cutter, S. L., Emrich, C. T., & Zêzere, J. L. (2015). Application of Social Vulnerability Index (SoVI) and delineation of natural risk zones in Greater Lisbon, Portugal. *Journal of Risk Research*, 18(5), 651–674. https://doi.org/10.1080/13669877.2014.910689
- Hansen, F., & Smidt-Jensen, S. (2004). Shrinking vulnerability in the Danish periphery? the case of Nakskov. *Magazin Städte Im Umbruch*, 11, 23–26.
- Hauer, M. E., Evans, J. M., & Mishra, D. R. (2016). Millions projected to be at risk from sea-level rise in the continental United States. *Nature Climate Change*, 6(7), 691–695. https://doi.org/10.1038/nclimate2961
- Heinz, H. J., & Tufford, D. (2001). The hidden costs of coastal hazards; implications for risk assessment and mitigation. *Electronic Green Journal*.

Hewitt, K. (1983). The idea of calamity in a technocratic age. In Interpretations of Calamity.

- Hewitt, K. (1997). *Regions of Risk : a Geographical Introduction to Disasters. Themes in resource management.* https://doi.org/10.1016/S0143-6228(97)00049-0
- Holand, I. S., Lujala, P., & Rod, J. K. (2011). Social vulnerability assessment for Norway: A quantitative approach. *Norsk Geografisk Tidsskrift*, 65(1), 1–17. https://doi.org/10.1080/00291951.2010.550167
- Hummell, M. B. de L., Cutter, S. L., & Emrich, C. T. (2016). Social Vulnerability to Natural Hazards in Brazil. *International Journal of Disaster Risk Science*, 7(2), 111–122. https://doi.org/10.1007/s13753-016-0090-9
- Intergovernmental Panel on Climate Change (IPCC. (2007). Fourth Assessment Report (AR4) of the Intergovernmental Panel on Climate Change. Cambridge.
- IPCC. (2001a). Climate change: impacts, adaptation, and vulner- ability. Contribution of Working Group II to the third assess- ment report of the Intergovernmental Panel on Climate Change. *Cambridge University Press*.
- IPCC. (2001b). Climate Change 2001: The Scientific Basis. Cambridge Uniersity Press. https://doi.org/10.1256/004316502320517344
- Jebens, M, Sorensen, C., & Piontkowitz, T. (2016). Danish risk management plans of the EU Floods Directive. *E3S Web of Conferences*, 7. https://doi.org/10.1051/e3sconf/20160723005
- Jebens, M., & Sorensen, C. (2017). Coping with Future Coastal Floods in Denmark—Advancing the Use of Global Frameworks, 239–259. https://doi.org/10.1007/978-3-319-70703-7\_13
- Joakim, E. (2017). Assessing the 'Hazards of Place 'Model of Vulnerability: A Case Study of Waterloo Region Assessing the 'Hazards of Place 'Model of Vulnerability: A Case Study of Waterloo Region By: Submitted to the Department of Geography and Environmental Studies, (October).
- Kaperson, J. X., & Kasperson, R. E. (2001). International Workshop on Vulnerability and Global Environmental Change, 17–19 May 2001: A Workshop Summary. Stockholm.
- Kasperson, R. E., Kasperson, J. X., Turner, B. L., Hsieh, W., & Schiller, A. (2000). Vulnerability to Global Environmental Change, The Human Dimensions of Global Environmental Change. Cambridge.: MIT Press,.
- Kelly, P. M., & Adger, W. N. (2000). Theory and Practice in Assessing Vulnerability To. *Climatic Change*, 47, 325–352. https://doi.org/10.1158/0008-5472.CAN-11-2495
- Kuhlicke, C., Scolobig, A., Tapsell, S., Steinführer, A., & de Marchi, B. (2011). Contextualizing social vulnerability: Findings from case studies across Europe. *Natural Hazards*, 58(2), 789–810. https://doi.org/10.1007/s11069-011-9751-6
- Kundzewicz, Z. W. (2012). Changes in Flood Risk in Europe. (Zbigniew W. Kundzewicz, Ed.), Changes in Flood Risk in Europe. IAHS Press. https://doi.org/10.1201/b12348
- Kundzewicz, Z. W., Kanae, S., Seneviratne, S. I., Handmer, J., Nicholls, N., Peduzzi, P., ... Sherstyukov, B. (2014). Flood risk and climate change: global and regional perspectives. *Hydrological Sciences Journal*. https://doi.org/10.1080/02626667.2013.857411

- Kystdirektoratet. (2018a). Metode til national risikovurdering af oversvømmelse fra hav og vandløb samt ajourføring af risikoområder Oversvømmelsesdirektivet Anden planperiode FORSLAG. Retrieved from www.kyst.dk
- Kystdirektoratet. (2018b). Risikoområder i Danmark. Retrieved March 4, 2019, from https://oversvommelse.kyst.dk/
- Laursen, L. L. H. (2009). SHRINKING CITIES OR URBAN TRANSFORMATION ! PhD-thesis submitt ed for assessment at the Doctoral School of Planning and.
- Lipinsky, B., & McGray, H. (2010). Summary of Studies Estimating the Cost of Climate Change Adaptation in the Developing World, (January 2010), 2006–2008.
- Marchand, M. (2009). Modelling Coastal Vulnerability: Design and Evaluation of a Vulnerability Model for Tropical Storms and Floods. IOS Press (Vol. 5). https://doi.org/10.1007/s13398-014-0173-7.2
- McGranahan, G., Balk, D., & Anderson, B. (2007). The rising tide: Assessing the risks of climate change and human settlements in low elevation coastal zones. *Environment and Urbanization*, 19(1), 17–37. https://doi.org/10.1177/0956247807076960
- McLaughlin, S., Andrew, J., & Cooper, G. (2010). A multi-scale coastal vulnerability index: A tool for coastal managers? *Environmental Hazards*, 9(3), 233–248. https://doi.org/10.3763/ehaz.2010.0052
- Mileti D.S. (1999). Disasters by design: a reassessment of natural hazards in the United States, Natural hazards and disasters. *Joseph Henry Press*. https://doi.org/10.1016/j.gloenvcha.2008.07.013
- Ministry of the Environment and Food of Denmark, & Environmental Protection Agency. (n.d.). Climate Change<br/>Adaptation.RetrievedMarch4,2019,fromhttps://en.klimatilpasning.dk/knowledge/climate/denmarksfutureclimate666666
- Moore, H. F. (2010). Immigration in Denmark and Norway, 82(3), 355–364.
- Morrow, B H, & Peacock, W. G. (1997). Disasters and social change: Hurricane Andrew and the reshaping of Miami? *Hurricane Andrew: Ethnicity, Gender and the Sociology of Disasters*.
- Morrow, B. H. (1999). Identifying and mapping community vulnerability. *Disasters*. https://doi.org/10.1111/1467-7717.00102.
- Mostert, E., & Junier, S. J. (2010). The European flood risk directive: challenges for research. *Hydrology and Earth System Sciences Discussions*, 6(4), 4961–4988. https://doi.org/10.5194/hessd-6-4961-2009
- Myers, C. A., Slack, T., & Singelmann, J. (2008). Social vulnerability and migration in the wake of disaster: The case of Hurricanes Katrina and Rita. *Population and Environment*, 29(6), 271–291. https://doi.org/10.1007/s11111-008-0072-y
- Nerem, R. S., Beckley, B. D., Fasullo, J. T., Hamlington, B. D., Masters, D., & Mitchum, G. T. (2018). Climatechange–driven accelerated sea-level rise detected in the altimeter era. *Proceedings of the National Academy of Sciences*, 115(9), 2022–2025. https://doi.org/10.1073/pnas.1717312115
- Neumann, B., Vafeidis, A. T., Zimmermann, J., & Nicholls, R. J. (2015). Future coastal population growth and exposure to sea-level rise and coastal flooding - A global assessment. *PLoS ONE*. https://doi.org/10.1371/journal.pone.0118571

- Nicholls, R. J., Hoozemans, F. M. J., & Marchand, M. (1999). Increasing flood risk and wetland losses due to global sea-level rise: Regional and global analyses. In *Global Environmental Change*. https://doi.org/10.1016/S0959-3780(99)00019-9
- O'Brien, K. L., Eriksen, S., Schjolden, A., & Nygaard, L. (2004). What's in a word? Conflicting interpretations of vulnerability in climate change research. *CICERO Working Paper*, 04(2004:04), 16. Retrieved from http://dspace.cigilibrary.org/jspui/handle/123456789/7304
- O'Brien, P., & Mileti, D. (1992). Citizen participation in emergency response following the Loma Prieta earthquake.pdf. *International Journal of Mass Emergencies and Disasters*.
- O'Keefe, P., Westgate, K., & Wisner, B. (1976). Taking the naturalness out of natural disasters. *Nature*, 260(5552), 566–567. https://doi.org/10.1038/260566a0
- Oreskes, N. (2004). The Scientific Consensus on Climate Change. *Science*, 306(5702), 1686–1686. https://doi.org/10.1126/science.1103618
- Peacock, W G, Morrow, B. H., & Gladwin, H. (1997). Hurricane Andrew and the Reshaping of Miami: Ethnicity, Gender, and the Socio-Political Ecology of Disasters. (W G Peacock, B. H. Morrow, & H. Gladwin, Eds.). Gainesville, Fla.: University Press of Florida.
- Peduzzi, P., Sherstyukov, B., Kundzewicz, Z. W., Muir-Wood, R., Mechler, R., Nicholls, N., ... Bouwer, L. M. (2013). Flood risk and climate change: global and regional perspectives. *Hydrological Sciences Journal*, 59(1), 1–28. https://doi.org/10.1080/02626667.2013.857411
- Pelling, M. (2003). Social Capital, Hazards and Adaptation Strategies for the Vulnerable. Draft. Jus- tice in Adaptation to Climate Change.
- Platt, R. (1999). *Disasters and Democracy: The Politics of Extreme Natural Events*. Washington D.C.: Island References Press.
- Puente, S. (1999). Social Vulnerability to Disaster in Mexico City. In J. Mitchell (Ed.), Crucibles of Hazard: Mega-Cities and Disasters in Transition. (pp. 295–334.). Tokyo: United Nations University Press.
- Ritchie, J., Lewis, J., Nicholls, C. M., & Ormston, R. (2013). The foundation of qualitative research. *Qualitative Research Practice: A Guide for Social Science Students and Researchers*.
- Rodda, H., Plate, E., Schellnhuber, H., Strupczewski, W., & Rodda, J. (2012). Changes in Flood Risk Setting the Stage. In *Changes in Flood Risk in Europe*. https://doi.org/10.1201/b12348-4
- Rufat, S., Tate, E., Burton, C. G., & Maroof, A. S. (2015). Social vulnerability to floods: Review of case studies and implications for measurement. *International Journal of Disaster Risk Reduction*, 14, 470–486. https://doi.org/10.1016/j.ijdrr.2015.09.013
- Russell, B. H. (2006). Methods in Anthropology Qualitative and Quantative.
- Rygel, L., O'Sullivan, D., & Yarnal, B. (2006). A method for constructing a social vulnerability index: An application to hurricane storm surges in a developed country. *Mitigation and Adaptation Strategies for Global Change*, 11(3), 741–764. https://doi.org/10.1007/s11027-006-0265-6
- Schellnhuber, H., Serdeczny, O., Adams, S., Kohler, C., Otto, I. M., & Schleussner, C. F. (2016). The challenge of a 4?C World by 2100. In H. . Brauch (Ed.), *Hexagon series on human environmental security and*

peace. Springer.

- Scheuer, S., Haase, D., & Meyer, V. (2011). Exploring multicriteria flood vulnerability by integrating economic, social and ecological dimensions of flood risk and coping capaciy: from a starting point view towards an end point view of vulnerability. *Natural Hazards*, (58), 731–751.
- Schumann, L. R. M. A., & Moura, L. B. A. (2015). vulnerability synthetic indices: a literature integrative review. *Ciência & Saúde Coletiva*, 20(7), 2105–2120. https://doi.org/10.1590/1413-81232015207.10742014
- Singh, K. (2014). Application of Pressure and Release (PAR) Model for Assessing Vulnerability to Industrial Hazards in District Bathinda (Punjab, India). *International Journal of Management and Social Sciences Research (IJMSSR)*, *3*(5), 2319–4421. https://doi.org/10.1016/j.envpol.2015.08.042
- Singh, S. R., Eghdami, M. R., & Singh, S. (2014). The Concept of Social Vulnerability: A Review from Disasters Perspectives. *International Journal of Interdisciplinary and Multidisciplinary Studies*, 1(6), 71– 82. Retrieved from http://www.ijims.com
- Sorensen, C., & Jebens, M. (2015). How can awareness in civil society and in governance be raised? Reducing risks from coastal hazards. In Proceedings of the 24th New South Wales Coastal Conference.
- Sorensen, C., Knudsen, P., Sorensen, P., Damgaard, T., Molgaard, M. R., & Jensen, J. (2018). Rethinking Coastal Community Approaches to Climate Change Impacts and Adaptation. *Journal of Coastal Research*, 85(85), 1521–1525. https://doi.org/10.2112/si85-305.1
- Stormrådet. (2009). Stormflods skadestatistik maj 2009.
- Sun, Y., Solomon, S., Dai, A., & Portmann, R. W. (2006). How often does it rain? *Journal of Climate*. https://doi.org/10.1175/JCLI3672.1
- Turner, B. L., Christensen, L., Matson, P. A., Eckley, N., Martello, M. L., Schiller, A., ... Polsky, C. (2003). A framework for vulnerability analysis in sustainability science. *Proceedings of the National Academy of Sciences*, 100(14), 8074–8079. https://doi.org/10.1073/pnas.1231335100
- U.S. Global Change Research Program. (2017). Executive summary. In: Climate Science Special Report: Fourth National Climate Assessment, Volume I, 10–34. https://doi.org/10.7930/J0DJ5CTG.U.S.
- UNDP. (2004). REDUCING DISASTER RISK: A CHALLENGE FOR DEVELOPMENT-A GLOBAL REPORT. Disaster and Crisis Management.
- UNDRO. (1991). Mitigating natural disasters : Phenomena, effects and options : A manual for policy makers and planners.
- UNEP. (2002). Global Environment Outlook (GEO 3). Past, present and future perspectives. United Nations Environment Programme.
- UNEP. (2015). The Adaptation Finance Gap Update: With Insights from the INDCs. Retrieved from http://apps.unep.org/publications/index.php?option=com\_pub&task=download&file=011914\_en
- UNISDR. (2004). Living with risk: a global review of disaster reduction initiatives. Geneva: UN International Strategy for Disaster Reduction. https://doi.org/9211010640

United Nations. (1993). Living with Risk - Vol II (Vol. II).

- Villagran De Leon, J. C. (2006). Vulnerability: a conceptual and methodological review. UNU Intstitute for Environment and Human Security. https://doi.org/10.1017/CBO9781107415324.004
- Vinod, T. (2017). CLimate Change and Natural Disasters.
- Walliman, N. (2011a). Qualitative data analysis. *Social Research Methods*, 46. https://doi.org/https://dx.doi.org/10.4135/9781849209939
- Walliman, N. (2011b). Quantitative Data Analysis. Social Research Methods, 110–128. https://doi.org/10.4135/9781849209939.n10
- Watts, M. J., & Bohle, H. G. (1993). The space of vulnerability: The causal structure of hunger and famine. *Progress in Human Geography*, *17*(1), 43–67. https://doi.org/10.1177/030913259301700103
- White, G. F., & Haas, J. E. (1975). Assessment of Research on Natural Hazards. (M. Cambridge & M. Press., Eds.).
- White, P., Pelling, M., Sen, K., Seddon, D., Russel, S., & Few, R. (2005). *Disaster risk reduction: a development concern. A scoping study on links between disaster risk reduction, poverty and development.* Department for International Development.
- Willis, I., & Fitton, J. (2016). A review of multivariate social vulnerability methodologies: A case study of the River Parrett catchment, UK. *Natural Hazards and Earth System Sciences*, 16(6), 1387–1399. https://doi.org/10.5194/nhess-16-1387-2016
- Willis, I., Gibin, M., Barros, J., & Webber, R. (2010). Applying neighbourhood classification systems to natural hazards: A case study of Mt Vesuvius. *Natural Hazards*, 70(1), 1–22. https://doi.org/10.1007/s11069-010-9648-9
- Wisner, B., Blaikie, P., Cannon, T., & Davis, I. (2004). At Risk Natural Hazards, People's Vulnerability and Disasters second edition: Routledge, London. At Risk: Natural Hazards, People's Vulnerability and Disasters. https://doi.org/https://doi.org/10.2202/1547-7355.1131
- Wren, K. (2001). Cultural racism: Something rotten in the state of Denmark? *Social and Cultural Geography*, 2(2), 141–162. https://doi.org/10.1080/14649360120047788
- Wuebbles, D. J., Fahey, D. W., Hibbard, K. A., Dokken, D. J., Stewart, B. C., & Maycock, T. K. (Eds.). (2017). *Climate Science Special Report: Fourth National Climate Assessment, Volume I.* Washington, DC. https://doi.org/10.7930/J0J964J6
- Yelvington, K. (1997). Coping in a temporary way: The tent cities. In Walter Gillis Peacock, B. H. Morrow, & H. Gladwin (Eds.), *Hurricane Andrew: Ethnicity, Gender, and the Sociology of Disasters* (pp. 92–115). Miami, FL: International Hurricane Center, Laboratory for Social and Behavioral Research.